SECURITIES AND EXCHANGE COMMISSION

# FORM 8-K

Current report filing

Filing Date: 2022-06-21 | Period of Report: 2022-06-14 SEC Accession No. 0001654954-22-008585

(HTML Version on secdatabase.com)

# **FILER**

# Zomedica Corp.

CIK:1684144| IRS No.: 000000000 | State of Incorp.:A0 | Fiscal Year End: 1231 Type: 8-K | Act: 34 | File No.: 001-38298 | Film No.: 221028330 SIC: 2834 Pharmaceutical preparations Mailing Address 100 PHOENIX DRIVE SUITE 125 ANN ARBOR MI 48108 Business Address 100 PHOENIX DRIVE SUITE 125 ANN ARBOR MI 48108 (734) 369-2555

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

# FORM 8-K

# **CURRENT REPORT**

Pursuant to Section 13 or 15(d)

of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): June 14, 2022

# ZOMEDICA CORP.

(Exact name of registrant as specified in its charter)

Alberta, Canada

001-38298

(State or Other Jurisdiction of Incorporation)

(Commission File Number)

N/A (I.R.S. Employer Identification No.)

100 Phoenix Drive, Suite 125 Ann Arbor, Michigan 48108

(Address of Principal Executive Offices) (Zip Code)

(734) 369-2555

(Registrant's telephone number, including area code)

Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Shares, without par value	ZOM	NYSE American

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company  $\Box$ 

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.  $\Box$ 

#### Item 1.01. Entry into a Material Definitive Agreement.

On June 14, 2022, Zomedica Corp. (the "Company") and its wholly-owned subsidiary Zomedica Inc. entered into an Asset Purchase Agreement (the "Purchase Agreement") with Revo Squared LLC (the "Revo Squared") and its majority member (Revo Squared and it majority member are referred to herein as the "Selling Parties") pursuant to which Zomedica Inc. agreed to acquire substantially all of the assets of Revo Squared (the "Acquisition"). Revo Squared, based in Marietta, Georgia, is in the business of developing, manufacturing, marketing, distributing and selling diagnostic imaging products and services for use in animal health, including its SuperView, Sonoview Color, Sonoview Mini/Mini Plus and Microview product offerings (the "Revo Squared Products").

The base purchase price for the Acquisition will be \$6.0 million in cash, which will be subject to adjustments based on the amount of Revo's Squared's working capital at the closing of the Acquisition. \$500,000 of the base purchase price will be deposited into a thirdparty escrow account for a period of 15 months to support Revo Squared's indemnification obligation under the Purchase Agreement. The Company has also agreed to issue ten-year warrants (the "Warrants") at the closing of the Acquisition to purchase an aggregate of 10,000,000 of the Company's common shares to Revo Squared at a per share exercise price equal to the closing price of the Company's common shares either on the trading day immediately prior to the date of closing of the Acquisition (if the closing occurs prior to the close of trading on the date of closing) or on the date of the closing of the Acquisition (if the closing occurs after the close of trading). Zomedica Inc. has also agreed to pay Revo Squared aggregate earn-out payments of up to \$4.0 million based on the achievement of milestones related to future net sales from Revo Squared Products. One-time earn-out payments of \$2.0 million each will be payable upon net sales from Revo Squared Products exceeding \$5.0 million and \$10.0 million during any calendar year ending on or prior to December 31, 2027.

The Purchase Agreement contains customary representations, warranties and covenants of the parties. The Purchase Agreement also contained indemnification provisions pursuant to which the Selling Parties have agreed to indemnify Zomedica Inc. and its affiliate against certain losses, subject to the limitations set forth therein, including losses related to breaches of representations, warranties and covenants.

The closing of the Acquisition is subject to closing conditions set forth in the Purchase Agreement. Consummation of the Acquisition is expected to occur on or around July 1, 2022, subject to the satisfaction or waiver of closing conditions.

The Purchase Agreement is attached to this report as Exhibit 2.1 and incorporated herein by reference. The foregoing description of the Purchase Agreement and the transactions contemplated thereby is not complete and is qualified in its entirety by the contents of the actual Purchase Agreement.

The Purchase Agreement is included to provide investors and security holders with information regarding its terms. It is not intended to provide any other factual information about the Revo Squared or the Company or any of their respective businesses, subsidiaries or affiliates. The representations, warranties and covenants contained in the Purchase Agreement were made by the parties thereto only for purposes of the Purchase Agreement and as of specific dates, and were made solely for the benefit of the parties to the Purchase Agreement. Investors should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts or condition of the Revo Squared or the Company or any of their respective subsidiaries or affiliates.

#### Item 8.01. Other Events.

On June 15, 2022, the Company issued a press release announcing entry into the Purchase Agreement. A copy of the press release is attached hereto as Exhibit 99.1 and incorporated herein by reference.

#### Item 9.01. Financial Statements and Exhibits.

(a) Financial statements of businesses acquired

If required, and in accordance with Item 9.01(a)(3) of Form 8-K, the financial statements required by Item 9.01(a) of Form 8-K will be filed as necessary by amendment to this Form 8-K within 71 calendar days after the required filing date for this Current Report.

#### (b) Pro forma financial information

If required, and in accordance with Item 9.01(b)(2) of Form 8-K, the financial information required by Item 9.01(b) will be filed as necessary by amendment to this Form 8-K within 71 calendar days after the required filing date for this Current Report.

(d)	Exhibit No.	Description
	<u>2.1*</u>	Asset Purchase Agreement dated June 14, 2022 by and among Revo Squared LLC, the Principal Member (as defined therein), Zomedica Inc. and Zomedica Corp.
	<u>99.1</u>	Press Release, dated June 15, 2022
	104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

\* Certain exhibits and/or schedules to this Exhibit have been omitted in accordance with Regulation S-K Item 601(a)(5). The Company agrees to furnish a supplemental copy of any omitted exhibit or schedule to the SEC upon its request.

3

#### SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

# ZOMEDICA CORP.

Date: June 21, 2022

#### By: /s/Ann Marie Cotter Ann Marie Cotter Chief Financial Officer

4

# EXHIBIT INDEX

(d)	Exhibit No.	Description
	<u>2.1</u>	Asset Purchase Agreement dated June 14, 2022 by and among Revo Squared LLC, the Principal Member (as defined therein), Zomedica Inc. and Zomedica Corp.
	<u>99.1</u>	Press Release, dated June 15, 2022
	104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

5

# ASSET PURCHASE AGREEMENT

#### BETWEEN

### **REVO SQUARED LLC,**

# THE PRINCIPAL MEMBER OF REVO SQUARED LLC (AS DEFINED HEREIN),

# ZOMEDICA INC.

AND

# ZOMEDICA CORP.

June 14, 2022

# TABLE OF CONTENTS

1	ARTICLE	I. DEFINITIONS	
	1.1	Certain Definitions	1
	1.2	Interpretation	9
1	ARTICLE	II. PURCHASE AND SALE OF THE ACQUIRED ASSETS	10
	2.1	Purchase and Sale of the Acquired Assets	10
	2.2	Excluded Assets	11
	2.3	Assumption of Certain Liabilities	11
	2.4	Nontransferable Assets	12
	2.5	Closing Consideration	12
	2.6	Determination of the Final Closing Date Purchase Price Closing Payment	13
	2.7	Closing	14
	2.8	Closing Deliveries.	14
	2.9	Earnout Payments	16
	2.10	Withholding	17
1	ARTICLE	III. REPRESENTATIONS AND WARRANTIES OF SELLER AND PRINCIPAL MEMBER	17
	3.1	Organization	17
	3.2	Ownership	17
	3.3	Authority, Validity and Enforceability	17
	3.4	No Subsidiaries	18
	3.5	No Conflict	18

3.6	Consents	18
3.7	Financial Statements; Undisclosed Liabilities.	18
3.8	Accounts Receivable	19
3.9	Absence of Certain Developments	19
3.10	Compliance with Laws; Governmental Authorizations; Licenses	19
3.11	Litigation	20
3.12	Real Property; Person Property.	20
3.13	Taxes.	21
3.14	Environmental Matters	22
3.15	Employee Matters	22
3.16	Employee Benefit Plans.	22
3.17	Intellectual Property Rights	23
3.18	Material Contracts	25
3.19	Insurance	27
3.20	Title to Assets	27
3.21	Customers and Suppliers.	27
3.22	Affiliate Transactions	28
3.23	Brokers	28
3.24	Product Liability Claims	28
3.25	Warranties	28
3.26	Securities Law	28
3.27	Disclosure	29
3.28	No Other Representations	29

# TABLE OF CONTENTS

(continued)

			Page
AR'	TICLE	IV. REPRESENTATIONS AND WARRANTIES OF THE BUYER PARTIES	30
	4.1	Organization	30
	4.2	Authority, Validity and Enforceability	30
	4.3	No Conflict	30
	4.4	Consents	30
	4.5	Issuance of the Securities	31
	4.6	Brokers	31
	4.7	Solvency	31
	4.8	Independent Investigation	31
	4.9	No Other Representations	31
AR'	TICLE	C V. COVENANTS AND AGREEMENTS	32
	5.1	Access and Information.	32
	5.2	Conduct of Business Prior to Closing	32
	5.3	Commercially Reasonable Efforts; Notification of Certain Events.	33
	5.4	Public Announcements	33
	5.5	Employee Matters.	33
	5.6	Exclusivity	34
	5.7	Tax Matters.	35
	5.8	Wrong Pockets; Transfer of Certain Funds Received Post-Closing	36
	5.9	Post-Closing Transition Services	36
	5.10	Further Assurances	36
AR	TICLE	E VI. CONDITIONS TO CLOSING	37
	6.1	Mutual Conditions	37
	6.2	Conditions to the Obligations of the Buyer	37
	6.3	Conditions to the Obligations of Seller Parties	37
AR'	TICLE	CVII. TERMINATION	38

	7 1		20	
	7.1	Termination	38	
	7.2	Effect of Termination	38	
ARTICLE VIII. SURVIVAL; INDEMNIFICATION				
	8.1	Survival of Representations, Warranties and Covenants	39	
	8.2	Indemnification.	39	
	8.3	Limitations on Liability	40	
	8.4	Claims Procedures	41	
	8.5	Payment of Claim	42	
	8.6	Exclusive Remedy	43	
	8.7	Calculation of Loss	43	
	8.8	Investigation	43	
	8.9	Treatment of Indemnity Payments	43	
	8.10	Schedules	43	
A	RTICLE	IX. MISCELLANEOUS	44	
	9.1	Notices	44	
	9.2	Computation of Time	44	
	9.3	Expenses	44	

-ii-

#### TABLE OF CONTENTS (continued)

		Page
9.4	Governing Law; Jurisdiction	44
9.5	Parties in Interest	45
9.6	Counterparts	45
9.7	Entire Agreement; Amendment; Waiver; Assignment	45
9.8	Severability	45
9.9	No Strict Construction	45
9.10	Waiver of Jury Trial	46
9.11	Specific Performance	46
9.12	Failure or Indulgence not Waiver	

-iii-

#### ASSET PURCHASE AGREEMENT

THIS ASSET PURCHASE AGREEMENT (this "<u>Agreement</u>"), dated \_\_\_\_\_\_, 2022 (the "<u>Effective Date</u>"), is by and between Revo Squared LLC, a Georgia limited liability company ("<u>Seller</u>"), the Principal Member (as defined herein), Zomedica Inc., a Delaware corporation ("<u>Buyer</u>"), and Zomedica Corp., an Alberta corporation ("<u>Buyer Parent</u>"). Seller and the Principal Member are collectively referred to herein as the "<u>Seller Parties</u>"). Buyer and Buyer Parent are collectively referred to herein as the "<u>Buyer Parties</u>"). The Buyer Parties and the Seller Parties are each referred to as a "<u>Party</u>" and, collectively, as the "<u>Parties</u>."

# RECITALS

- A. Seller is in the business of developing, manufacturing, marketing, distributing and selling diagnostic imaging products and services for use in animal health, including its SuperView, Sonoview Color, Sonoview Mini/Mini Plus and Microview product offerings (collectively, the "<u>Business</u>").
- B. The Principal Member owns the majority of the outstanding membership interests of Seller, and will benefit financially from the transactions contemplated by this Agreement.

- C. Buyer is a wholly-owned subsidiary of Buyer Parent.
- D. Seller wishes to sell, transfer, convey and assign to Buyer, and the Buyer Parties wish for Buyer to acquire and assume, the substantially all of the assets comprising or pertaining to the Business, upon the terms and subject to the conditions set forth herein.

NOW, THEREFORE, in consideration of the mutual covenants contained herein, the Parties agree as follows:

### ARTICLE I. DEFINITIONS

1.1<u>Certain Definitions</u>. As used in this Agreement, the following terms have the respective meanings set forth below.

"<u>Action</u>" means any suit, litigation, action, claim, appeal, inquiry, investigation, cause of action, arbitration or other similar legal proceeding (whether in contract, tort or otherwise, whether at law or in equity and whether civil or criminal) by or before a Governmental Authority.

"<u>Acquisition Proposal</u>" means, other than the transactions contemplated hereby, any inquiry, proposal, indication of interest or offer (whether written or oral) with respect to any direct or indirect: (a) purchase or sale of an Equity Interest (including by means of a tender or exchange offer) representing more than twenty-five percent (25%) of the voting power in Seller; (b) merger, consolidation, other business combination, reorganization, recapitalization, share exchange, dissolution, liquidation or similar transaction involving Seller; (c) purchase or sale of assets, businesses, securities or ownership interests representing more than twenty-five percent (25%) of the net revenues, net income or net assets of the Business, taken as a whole, or of Seller, taken as a whole. Notwithstanding the foregoing, discussions and dealings with the holders of Indebtedness with respect to satisfying such Indebtedness by Seller shall not constitute an Acquisition Proposal, as long as such dealings do not involve the transfer of any of the Acquired Assets to such holders of Indebtedness.

"Adjustment Escrow Amount" means an amount equal to \$50,000.00.

"<u>Affiliate</u>" means, with respect to any Person, any Person directly or indirectly controlling, controlled by, or under common control with, such other Person as of the date on which, or at any time during the period for which, the determination of affiliation is being made. The term "<u>control</u>" (including the correlative meanings of the terms "<u>controlled by</u>" and "<u>under common control with</u>"), as used with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management policies of such Person, whether through the ownership of Equity Interests, by Contract or otherwise.

"<u>Anti-corruption Laws</u>" means any Legal Requirements relating to anti-bribery or anti-corruption (governmental or commercial) that apply to Seller or the Business, including Legal Requirements that prohibit the corrupt payment, offer, promise or authorization of the payment or transfer of anything of value (including gifts or entertainment), directly or indirectly, to any foreign Governmental Authority, foreign Government Official, foreign government employee, state-owned entity employee or commercial entity to obtain or retain a business advantage.

"Base Purchase Price" means \$6,000,000.00.

"Bill of Sale," means the Bill of Sale, Assignment and Assumption Agreement to be executed at Closing by Seller and Buyer.

"Business Day" shall mean any day except a Saturday, a Sunday or any other day on which commercial banks are required or authorized to close in New York, New York.

"Business Intellectual Property Rights" means all Owned Intellectual Property and all Licensed Intellectual Property.

"<u>Cash</u>" means, as of the Closing, all cash and cash equivalents (including marketable securities, foreign exchange contracts, interest hedge facilities and deposits in transit) of Seller. Cash shall (i) include the amount of all undeposited funds (including deposits in transit from credit card companies), cash on hand, cash in registers, and checks and drafts issued to Seller which have been received by

Seller, in each case, as of immediately prior to Closing, and (ii) not include the amount of cash or cash equivalents in respect of checks or drafts issued by Seller prior to Closing to the extent such checks or drafts have not cleared as of immediately prior to Closing.

"<u>Claim Notice</u>" means written notification which contains (i) a description of the Loss incurred or reasonably expected to be incurred by the Indemnified Party, a reasonably specific description of the basis therefor and the claimed amount of such Loss incurred or reasonably expected to be incurred by the Indemnified Party, to the extent then known, (ii) a statement that the Indemnified Party is entitled to indemnification under ARTICLE VIII for such Loss and a reasonable explanation of the basis therefor, and (iii) a demand for payment in the amount of such Loss.

"Closing Indebtedness" means the amount of Indebtedness outstanding as of immediately prior to the Closing.

"<u>Closing Date Purchase Price</u>" means an amount in cash calculated as follows: (a) the Base Purchase Price; <u>minus</u> (b) the aggregate amount of Closing Indebtedness to be repaid pursuant to Section 2.5; <u>minus</u> (c) the Escrow Amount, <u>minus</u> (d) any Working Capital Deficiency, <u>plus</u> (e) any Working Capital Surplus. For purposes of clarification, no form of Indebtedness may be deducted twice in determining the Closing Date Purchase Price.

"Code" means the Internal Revenue Code of 1986, as amended.

-2-

"<u>Confidentiality Agreement</u>" means the Confidential Non-Disclosure Agreement dated November 17, 2021 by and between Seller and Buyer.

"<u>Contract</u>" means, with respect to any Person, any legally binding contract, agreement, deed, mortgage, lease, sublease, license, sublicense or other legally enforceable commitment, promise, undertaking, arrangement or understanding, whether written or oral, to which or by which such Person is a party or is otherwise bound.

"Contemplated Transactions" means the transactions contemplated by this Agreement and the other Transaction Documents.

"<u>Controlled Group</u>" means any trade or business (whether or not incorporated) (a) under common control, within the meaning of Section 4001(b)(1) of ERISA, with Seller or (b) which together with Seller is treated as a single employer under Section 414(t) of the Code.

"Disclosure Schedule" means the final disclosure schedule to this Agreement that are being executed and delivered by Seller in connection with the execution and delivery of this Agreement.

"Employee" means any employee of Seller.

"<u>Environment</u>" means soil, land surface or subsurface strata, surface waters (including navigable waters and ocean waters), groundwaters, drinking water supply, stream sediments, ambient air (including indoor air), plant and animal life and any other environmental medium or natural resource.

"Environmental Laws" shall mean any Applicable Law relating to pollution, protection or cleanup of the Environment, including CERCLA, the Resources Conservation and Recovery Act of 1976, as amended ("<u>RCRA</u>"), the Environment Quality Act and any other federal, state, provincial, local and municipal or foreign legal requirements relating to: (a) a Release or the containment, removal, remediation, response, cleanup or abatement of a Hazardous Substance; (b) the manufacture, generation, formulation, processing, labeling, distribution, introduction into commerce, use, treatment, handling, storage, or transportation of a Hazardous Substance; (c) exposure of Persons, including employees, to a Hazardous Substance; (d) occupational safety or health matters; and (e) the physical structure or condition of a building, facility, fixture or other structure, including those relating to the management, use, storage, disposal, cleanup or removal of asbestos, asbestos-containing materials, polychlorinated biphenyls or any other Hazardous Substance.

"<u>Environmental Permits</u>" means all Permits required to be held or obtained by Seller or the Business pursuant to Environmental Laws.

"Equity Interests" means (a) any capital stock, partnership, limited liability company or membership interest, unit of participation or other similar interest (however designated) in any Person and (b) any option, warrant, purchase right, conversion right,

exchange right or other Contract that would entitle any other Person to acquire any such interest in any such Person referred to in clause (a) or otherwise entitle any other Person to share in the equity, profits, earnings, losses or gains of any such Person referred to in clause (a) (including stock appreciation, phantom stock, profit participation or other similar rights).

"<u>Escrow Agent</u>" means U.S. Bank, National Association, or if such bank is unable to serve as Escrow Agent upon the terms set forth in this Agreement and the Escrow Agreement, such other escrow agent reasonably acceptable to Seller and Buyer.

"<u>Escrow Agreement</u>" means the Escrow Agreement by and among Buyer, Seller and the Escrow Agent, to be entered into at Closing, in substantially the form attached as <u>Exhibit A</u> (but subject to such revisions as the Escrow Agent may require that are reasonably acceptable to Buyer and Seller), which will contain a term of fifteen (15) months after the Closing Date.

"Escrow Amount" means the sum of the Adjustment Escrow Amount and the Indemnity Escrow Amount.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, including the rules and regulations promulgated thereunder.

"Fundamental Representations" means the representations and warranties set forth in Section 3.1 (Organization), Section 3.2 (Ownership), Section 3.3 (Authority, Validity and Enforceability), Section 3.5 (No Conflict), Section 3.6 (Consents), Section 3.11 (Litigation), Section 3.13 (Taxes), Section 3.16 (Employee Benefit Plans), Section 3.17 (Intellectual Property Rights), Section 3.20 (Title to Assets), Section 3.23 (Brokers), Section 3.26 (Securities Laws), Section 4.1 (Organization), Section 4.2 (Authority, Validity and Enforceability), Section 4.3 (No Conflict), Section 4.4 (Consents) and Section 4.6 (Brokers).

"GAAP" means United States generally accepted accounting principles as in effect from time to time.

"<u>Government Official</u>" means (a) any official, officer, employee or representative of, or any Person acting in an official capacity for or on behalf of, any Governmental Authority, (b) any political party or party official or candidate for political office or (c) any Person owned, in whole or in part, or controlled by any Person described in the foregoing clauses (a) or (b) of this definition.

"Governmental Authority" means any national, federal, state, provincial, county, municipal or local government, foreign or domestic, or the government of any political subdivision of any of the foregoing, or any court, tribunal, arbitrator, arbitration panel, entity, authority, agency, ministry or other similar body exercising executive, legislative, judicial, regulatory or administrative authority or functions of or pertaining to government, including any authority or other quasi-governmental entity established to perform any of such functions.

"<u>Hazardous Substances</u>" shall mean any toxic substance or waste, pollutant, hazardous substance or waste, contaminant, special waste, dangerous good, industrial substance or waste, petroleum or petroleum-derived substance or waste, or any toxic or hazardous constituent or any such substance or waste, including any substance regulated under or defined by Environmental Laws.

"Indebtedness" means, as of any date, with respect to Seller, without duplication and to the extent not taken into account in the calculation of Working Capital, (a) the outstanding principal amount of, accrued and unpaid interest on and other payment obligations (including any prepayment premiums, termination fees, expenses, breakage costs or penalties due upon payment of such indebtedness at Closing or otherwise payable as a result of the consummation of the transactions contemplated by this Agreement) arising under any obligations of Seller for borrowed money, (b) all obligations of Seller evidenced by (or which customarily would be evidenced by) any note, bond, debenture or other debt security or instrument, (c) the aggregate face amount of all outstanding letters of credit, bankers' acceptances or similar instruments issued on behalf of Seller, but excluding any undrawn amounts, (d) all obligations of Seller under conditional sale or other title retention agreements relating to property or assets purchased by Seller; (e) all equipment financing and capital lease obligations of Seller under leases that have been or should be capitalized in accordance with GAAP, (f) all obligations secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any lien on property owned or acquired by Seller, whether or not the obligations secured thereby have been assumed (g) all obligations of any other Person for borrowed money secured by (or for which the holder of such obligation has an existing right, contingent or otherwise, to be secured by) any Lien on any Acquired Asset, (h) any obligations based upon, resulting from or arising out of any unfunded benefit obligation that would be required to be included as a Liability as of the Closing on the financial statements of Seller prepared in accordance with GAAP, (i) any off-balance sheet Liabilities, and (j) all accounts payable not included in the calculation of the Working Capital.

"Indemnity Escrow Amount" means an amount equal to \$500,000.

"Intellectual Property Rights" means the following: (a) patents and patent applications, reexaminations, extensions and counterparts claiming priority therefrom, and any other indicia of invention ownership issued or granted by any Governmental Authority, applications for any of the foregoing, including provisional, utility, design, priority and other applications, divisionals, and continuations (in whole or in part), reissues or re-examinations of any of the foregoing, and moral and economic rights of inventors in any of the foregoing; (b) computer Software and firmware, including data files, Source code, object code and software-related specifications and documentation; (c) copyrights, whether in published or unpublished works, which include literary works and any other original works of authorship fixed in any tangible medium of expression (in whatever form now or hereafter existing), databases, data collections and rights therein, mask work rights, rights to compilations, collective works and derivative works of any of the foregoing and moral and economic rights of authors or creators in any of the foregoing, registrations and applications for registration for any of the foregoing; (d) trademarks, trade names, service marks, certification marks, service names, brands, trade dress and logos and the goodwill associated therewith; and (e) domain names; and (f) unpublished inventions (whether patentable or not), discoveries, improvements, designs, models, formulae, recipes, compilations, diagrams, drawings, blueprints, mask works, devices, methods, techniques, processes, know how, instructions, configurations, prototypes, samples, specifications, technology, trade secrets, confidential information, proprietary information, customer lists, and technical information, and moral and economic rights of authors and inventors in any of the foregoing. "Intellectual Property Rights" includes in each case any registrations of, applications to register, and renewals and extensions of, any of the foregoing items listed in clauses (a)-(e) of this definition with or by any Governmental Authority in any jurisdiction and similar or equivalent rights to any of the foregoing items listed in clauses (a)-(e) recognized by any Governmental Authority.

"Inventory" means all inventory, finished goods, raw materials, work in progress, packaging, supplies, parts and other inventories of Seller.

"IP Assignments" means one or more assignments of the Owned Intellectual Property, as required by Buyer or applicable Legal Requirements, in forms sufficient to the reasonable satisfaction of Buyer. With respect to United States Patent No. US 8,605,860 B2 (Dec. 10, 2013) (MULTI RESOLUTION X-RAY IMAGE CAPTURE); United States Patent No. US 8,369,483 B2 (Feb. 5, 2013) (MULTI RESOLUTION X-RAY IMAGE CAPTURE); and United States Patent No. US 10,175,154 B2 (Jan. 8, 2019) (METHODS AND SYSTEMS FOR SLIDE PROCESSING), "IP Assignments" shall include (i) assignments of each such patent/application from William Eugene Campbell to Seller, and (ii) subsequent assignments of each such patent/application from Seller to Buyer.

"IRS" means the Internal Revenue Service.

"Key Employee" means each of William E. Campbell and Negin Ghassemian.

"Knowledge" means, when referring to the 'knowledge' of the Seller, Seller's Knowledge, or any similar phrase or qualification based on knowledge, the actual knowledge of the Key Employees, and the knowledge that each Key Employee would have reasonably obtained after making due and appropriate inquiry of current Employees with respect to the particular matter in question.

-5-

"Legal Requirement" means any statute, law, rule, regulation, code, ordinance, Order of any Governmental Authority and similar provisions having the force and effect of law.

"<u>Liabilities</u>" includes liabilities, debts or other commitments or obligations, guarantees or endorsements, whether known or unknown, direct or indirect, absolute, accrued, contingent, liquidated, unliquidated or otherwise, due or to become due or otherwise, and whether or not required to be reflected on a balance sheet prepared in accordance with GAAP.

"<u>Licensed Intellectual Property</u>" means those Intellectual Property Rights that Seller or any of its Affiliates has been granted rights to or in from a third party that are: (A) used in or otherwise reasonably necessary for the making, having made, use, selling, offering for sale, export, importation or other use of the Business' products and services; or (B) otherwise used in or necessary to the conduct of the Business other than commercial off-the-shelf software licensed by Seller that are used solely in connection with the Business's internal operations and that have not been modified or customized by or for the Business.

"<u>Liens</u>" means any and all liens, charges, security interests, claims, mortgages, pledges, encumbrances, deeds of trust, judgments, voting trusts and other restrictions on title or transfer (including any conditional sale or other title retention agreement or lease in the nature thereof).

"Loss" means any and all actual damages, losses, Liabilities, obligations, claims of any kind, interest or expenses (including reasonable attorneys' fees and expenses, but expressly excluding any exemplary or punitive damages, except for any such damages to the extent actually awarded and paid or payable to a third party in connection with a Third-Party Claim). As it relates to indemnification claims under ARTICLE VIII, for purposes of determining whether a breach has occurred and calculating the amount of Losses incurred by a party seeking indemnification hereunder arising out of or resulting from any breach of a representation, warranty, covenant or agreement contained herein, references to "Material Adverse Effect" or materiality (or other similar terms) shall be disregarded.

"<u>Material Adverse Effect</u>" means any change, effect, event, occurrence, circumstance, state of facts or development that, individually or in the aggregate, is or is reasonably likely to have a material adverse effect on (a) condition (financial or otherwise), results of operations of Seller or the Business, (b) the Acquired Assets or the Assumed Liabilities, (c) the ability of the Seller Parties to timely consummate the transactions contemplated by this Agreement, or (d) the validity or enforceability of the Transaction Documents or the rights or remedies of Buyer thereunder.

"<u>Net Sales</u>" means the invoiced customer sale price, excluding taxes, duties, shipping, handling, wiring fees, transaction costs, distributor rebates and similar incentives, and currency conversion fees.

"Neutral Accountant" means Carr Riggs & Ingram CPA and Advisors LLC.

"Order" means any order, writ, judgment, injunction, decree (including any consent decree or similar agreed order or judgment), stipulation, ruling, determination or award issued or entered by any Governmental Authority.

"Organizational Documents" means, with respect to any Person (other than an individual), the certificate or articles of incorporation or organization of such Person and any limited liability company, stockholder, operating or partnership agreement, by-laws or similar documents or agreements (including all amendments thereto) relating to the legal organization or governance of such Person.

-6-

"Owned Intellectual Property" means any and all Intellectual Property Rights owned or purported to be owned by Seller or any of its Affiliates and used or held for use in the conduct of the Business. The Owned Intellectual Property excludes the Excluded Intellectual Property.

"<u>Permit</u>" means, with respect to any Person, any franchise, permit, accreditation, certification, consent, certificate or other similar authorization issued by, or otherwise granted by, any Governmental Authority, or any application therefor, to which or by which such Person is or will be subject or bound or to which or by which any property, business or operation of such Person is subject or bound.

"<u>Permitted Liens</u>" means (a) mechanics', materialmens', carriers', workmens', repairmens', contractors' or other similar Liens arising or incurred in the ordinary course of business that are not delinquent and are not material in any respect to the Acquired Assets or the Business, (b) easements, rights-of-way, restrictions and other similar charges and encumbrances of record not interfering materially with the ordinary conduct of the Business or the use or occupancy, or value of the assets subject thereto, and (c) Liens for Taxes not yet due and payable or for Taxes being contested in good faith by appropriate proceedings described in the Disclosure Schedule for which collection or enforcement is stayed and for which adequate reserve has been made according to GAAP as part of the calculation of the Working Capital.

"Person" means an individual, partnership, corporation, limited liability company, joint stock company, unincorporated organization or association, trust, joint venture, association or other organization, whether or not a legal entity, or a Governmental Entity.

"<u>Personal Data</u>" means a natural person's name, street address, telephone number, e-mail address, date of birth, photograph, social security number or tax identification number, credit card number, bank information, or biometric identifiers or any other piece of information that, alone or in combination with other information, allows the identification of, or contact with, any natural Person.

"Principal Member" means William E. Campbell, who is a signatory to this Agreement.

"<u>Purchase Price</u>" means the sum of (i) the Base Purchase Price, <u>plus</u> (ii) the Earnout Payment (to the extent payable pursuant to Section 2.9).

"<u>Revo Squared Products</u>" means the (i) MicroView product offering (including any derivates thereof or any products using the MicroView patent), (ii) SuperView product offering, (iii) SonovView Color product offering, (iv) SonoView Mini product offering, and (v) SonoView Mini Plus product offering.

"Securities Act" means the Securities Act of 1933, as amended, including the rules and regulations promulgated thereunder.

"<u>Seller Non-competition Agreements</u>" means the Non-Competition, Non-Solicitation and Confidentiality Agreements, to be executed at Closing, in favor of Buyer by Seller and the Principal Member, in the forms respectively attached as <u>Exhibits B-1</u> and <u>B-2</u>.

"<u>Software</u>" shall mean any software (regardless of whether such software is owned by or licensed to Seller by a third party) contained or included in or provided with any Seller products or services or used in the development, manufacturing, maintenance, repair, support, testing or performance of any Seller products or services, including but not limited to the Diagnostic Software.

-7-

"Source Code" shall mean any source code, or any portion, aspect or segment of any source code, relating to any Owned Intellectual Property or Licensed Intellectual Property or otherwise used by Seller or any of its Affiliates, including the Software.

"Straddle Period" means any Taxable period that begins on or before the Closing Date and ends after the Closing Date.

"<u>Subsidiary</u>" means, with respect to any Person, any entity of which securities or other ownership interests having ordinary voting power to elect a majority of the board of directors or other Persons performing similar functions are at any time directly or indirectly owned by such Person.

"Tax" (and with the corresponding meaning "Taxes" and "Taxable") means (a) any U.S. federal, state or local, or non-U.S. income, franchise, commercial activity, profits, gross receipts, ad valorem, net worth, value added, sales, use, real or personal property, license, occupation, payroll, withholding, employment or unemployment, disability, escheat, lost or unclaimed property, environmental, excise, severance, stamp, registration, premium, windfall profits, customs duties, capital stock, social security (or similar), transfer, estimated, alternative or add-on minimum tax payable or other tax, governmental fee or other like assessment or charge of any kind, including any interest, penalties or additions to tax imposed in respect of the foregoing, in each case, whether or not disputed, (b) any Liability for the payment of any amounts of any of the foregoing types as a result of being a member of an affiliated, consolidated, combined or taken into account with reference to the Liability of any other Person, (c) any Liability for the payment of any amounts as a result of being a party to any tax sharing or allocation agreements or arrangements (whether or not written) or with respect to the payment of any amounts of any of the foregoing types as a result of being a party to any tax sharing or allocation agreements or arrangements (whether or not written) or with respect to the payment of any amounts of any of the foregoing types as a result obligation to indemnify any other Person, and (d) any Liability for the payment of any of the foregoing types as a successor, transferee or otherwise.

"<u>Tax Returns</u>" means all United States federal, state, local and foreign Tax returns, declarations, statements, reports, schedules, forms and information returns or other documents and any amendments thereto required to be filed with a Tax Authority.

"<u>Taxing Authority</u>" means any United States, federal, state, local, or any foreign, Governmental Authority responsible for the imposition, determination, enforcement, assessment or collection of any Tax.

"<u>Transaction Documents</u>" means, collectively, this Agreement, the Bill of Sale, the Escrow Agreement, the Warrant, the Seller Non-competition Agreements, the IP Assignments, the Assignment and Assumption of Lease, and all other agreements and documents entered into in connection with the transactions contemplated by this Agreement.

"<u>Transfer Taxes</u>" means any sales, use, stock transfer, value added, real property transfer, real property gains, transfer, stamp, registration, documentary, recording or similar duties or Taxes together with any interest thereon, penalties, fines, costs, fees, additions to Tax or additional amounts with respect thereto incurred in connection with the transactions contemplated by this Agreement.

Date.

"Transferred Employees" means those Employees who accept offers of employment from Buyer effective as of the Closing

"Treasury Regulations" means the U.S. Treasury Regulations promulgated under the Code.

"<u>Working Capital</u>" means, as of the Closing, the current assets of the Business (including, without limitation, Accounts Receivable, Inventory, and Prepaid Items), plus (but without duplication) the full amount of the security deposit related to the Leased Real Property that is included in the Acquired Assets pursuant to Section 2.1(i), less the current liabilities of the Business that are Assumed Liabilities, as calculated in accordance with GAAP. Notwithstanding the foregoing, "Working Capital" shall not include Cash, Closing Indebtedness paid on behalf of Seller by Buyer at Closing pursuant to Section 2.5(a)(i), and Indebtedness due to Seller from William Campbell and Negin Ghassemian.

-8-

"Working Capital Deficiency" means the amount, if any, by which the Working Capital is less than the Working Capital Target.

"<u>Working Capital Surplus</u>" means the amount, if any, by which the Working Capital is greater than the Working Capital Target.

"Working Capital Target" means 125,000.00.

1.2 Interpretation. Each Schedule in the Disclosure Schedule shall be deemed to qualify only (i) the corresponding Section of this Agreement or (ii) any other Section of this Agreement to which such disclosure makes express reference or to which the relevance of such document is reasonably apparent on its face. Notwithstanding anything to the contrary contained herein, no disclosure in the Disclosure Schedule shall be deemed adequate to disclose an exception to a representation or warranty made by any Person unless the disclosure identifies the exception with reasonable particularity and describes the relevant facts in reasonable detail. Without limiting the generality of the foregoing, the mere listing (or inclusion of a copy) of a document or other item in the Disclosure Schedule shall not be deemed adequate to disclose an exception to a representation or warranty made herein (unless the representation or warranty has to do with the existence of the document or other item itself). Disclosure of any fact or item in any Section of the Disclosure Schedule shall not be considered an admission by Seller that such item or fact (or any non-disclosed item or information of comparable or greater significance) represents a material exception or fact, event or circumstance or that such item has had or would reasonably be expected to have a Company Material Adverse Effect, that such item or fact will in fact exceed any applicable threshold limitation set forth in the Agreement and shall not be construed as an admission by Seller or any Company of any non-compliance with, or violation of, any third party rights (including to any intellectual property rights) or any Applicable Law of any Governmental Entity, such disclosures having been made solely for the purposes of creating exceptions to the representations made herein or of disclosing any information required to be disclosed under the Agreement. When a reference is made in this Agreement to an article, section, paragraph, clause, schedule or exhibit, such reference shall be deemed to be to this Agreement unless otherwise indicated. The text of all schedules is incorporated herein by reference. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." As used herein, words in the singular will be held to include the plural and vice versa (unless the context otherwise requires), words of one gender shall be held to include the other gender (or the neuter) as the context requires, and the terms "hereof," "herein," and "herewith" and words of similar import will, unless otherwise stated, be construed to refer to this Agreement as a whole and not to any particular provision of this Agreement. With regard to all dates and time periods set forth or referred to in this Agreement, time is of the essence. The phrase "made available to Buyer" or similar phrases as used in this Agreement shall mean that the subject documents were either posted to the online data room (the "Data Room") of Buyer or delivered to Buyer or its accountants, attorneys or other agents.

#### ARTICLE II. PURCHASE AND SALE OF THE ACQUIRED ASSETS

2.1 <u>Purchase and Sale of the Acquired Assets</u>. On the terms and subject to the conditions of this Agreement, at the Closing, Seller shall sell, assign, convey and transfer to Buyer all of Seller's right, title and interest in, to and under all of the business, properties,

assets, goodwill and rights of Seller of whatever kind or nature, real or personal, tangible or intangible, owned, leased or licensed to Seller andused, held for use, or intended to be used in operating or maintaining the Business, wherever located and whether now existing or hereafter acquired, other than the Excluded Assets (collectively, the "<u>Acquired Assets</u>"). The Acquired Assets include, but are not limited to, the following:

- (a) All Inventory;
- (b) all supplies and equipment, wherever located, used or held for use in manufacturing, testing, storing or handling of the Business products;
- (c) all computers, servers, supplies, equipment and other fixed assets of the Business, wherever located, including the items listed on <u>Schedule 2.1(c)</u>;
- (d) all accounts receivable of Seller (the "<u>Accounts Receivable</u>");
- (e) all supplier and customer lists and pricing information relating to the Business;
- (f) all Contracts (including purchase orders) to which Seller is a party to the extent specifically related to the Business or any Transferred Employee, including those Contracts listed on <u>Schedule 2.1(f)</u> (the "<u>Assumed Contracts</u>");
- (g) all Business Intellectual Property Rights, including Software, Source Code, domain names, telephone and facsimile numbers, email addresses and the other Intellectual Property Rights used in the Business, as set forth on <u>Schedule</u> <u>2.1(g)</u>, and the goodwill associated therewith;
- (h) all Permits related to the Business;
- (i) all prepaid items, backlogs, advances and deposits of the Business to third parties, including but not limited to those set forth on <u>Schedule 2.1(i)</u> to be delivered at Closing (the "<u>Prepaid Items</u>");
- (j) all of Seller's claims, causes of action, defenses and rights of offset or counterclaim against third parties relating to any Acquired Asset or any Assumed Liability, including unliquidated rights under manufacturers' or vendors' warranties;
- (k) all books and records relating to the Business, including all accounting records, quality records, product designs, design history files, manufacturing drawings and all technical, sales and promotional literature;
- all insurance benefits to the extent relating to claims arising out of events that occurred prior to Closing (if any) and associated with the Acquired Assets, including such rights and proceeds receivable or hereafter received under any insurance policy written prior to the Closing;
- (m) all right, title and interest of Seller in any real property and improvements, including those leasehold interests set forth on <u>Schedule (m)</u> (the "Leased Real Property"); and
- (n) all goodwill associated with the Business and the Acquired Assets.

-10-

2.2 Excluded Assets. Notwithstanding anything contained herein to the contrary, Seller shall retain and Buyer shall not acquire or assume and the Acquired Assets shall not include the following assets, properties and rights (collectively, the "Excluded Assets"):

- (a) all Cash and all of Seller's bank accounts, including those bank accounts used for the Business;
- (b) all minute books, corporate seals, organizational documents, stock registers and such other books and records of Seller as pertain to ownership, organization or existence of Seller, and all Tax Returns and the Seller's general ledger, accounting records, and original accounting date (including all accounting software and data);
- (c) all shares or other equity interests or securities, or equivalents thereof, in any entity;

- (d) all personnel files for former employees of Seller who are not Transferred Employees and the Seller's personnel records and any other records that the Seller is required by law to retain in its possession for all employees of Seller;
- (e) all Benefit Plans, all documentation and materials related thereto, and all assets attributable thereto;
- (f) all rights of Seller under the Transaction Documents;
- (g) all insurance policies of Seller and rights thereunder, including rights as beneficiary or owner, and insurance claims for which the Seller may seek recovery under any of its existing policies, not including those insurance benefits set forth in Section 2.1(l);
- (h) any right to receive mail and other communications addressed to the Seller (subject to the Seller's obligation to forward to Buyer any mail, communications or other notices relating to the Acquired Assets);
- (i) all contracts with insurers for insurance coverage of the Seller, including any premium refunds, insurance proceeds (not including proceeds related to insurance benefits set forth in Section 2.1(l)) or credits; and
- (j) those personal effects and other assets specifically described on <u>Schedule 2.2(j)</u>.

2.3 <u>Assumption of Certain Liabilities</u>. Subject to the terms and conditions of this Agreement, at the Closing, only to the extent such Liabilities arise and accrue on or after the Closing Date and are not the result of a breach of any representation, warranty or covenant contained in this Agreement or a breach prior to the Closing Date of any Assumed Contract, to the extent not previously performed or discharged, Buyer shall assume and agree to pay, perform and discharge when due only the Liabilities in respect of (i) the Assumed Contracts, and (ii) those accounts payable included in the calculation of the Working Capital (the "<u>Assumed Liabilities</u>"). No assumption by Buyer of any of the Assumed Liabilities shall relieve or be deemed to relieve Seller from any Liability under any Transaction Document with respect to any representations or warranties or covenants made by Seller to Buyer. Except for the Assumed Liabilities, Buyer shall not assume pursuant to this Agreement or the transactions contemplated hereby, and shall have no liability for, any Liabilities of Seller or any of its Affiliates, or any of its predecessors in interest, of any kind, character or description whatsoever ("<u>Retained Liabilities</u>"), all of which shall be retained by and continue to be Liabilities of Seller or its Affiliates, as applicable. Retained Liabilities include, but are not limited to, each Liability based in whole or in part on events or conditions occurring or existing in connection with, or arising out of, Seller or the Business as operated prior to the Closing Date, or the manufacture, ownership, possession, use, operation or sale or other disposition prior to the Closing Date of any of the Acquired Assets (or any other assets, properties, rights or interests associated, at any time prior to the Closing Date, with the Business).

-11-

2.4 Nontransferable Assets. Notwithstanding any other provision of this Agreement or any Transaction Document, to the extent that any of the Assumed Contracts, or any other Acquired Assets are not assignable or otherwise transferable to Buyer without the consent, approval or waiver of another party thereto or any third party (including any Governmental Authority), or if such assignment or transfer would constitute a breach thereof or a violation of any applicable Legal Requirement or agreement with any third party, then neither this Agreement nor such Transaction Document shall constitute an assignment or transfer (or an attempted assignment or transfer) thereof until such consent, approval or waiver of such party or parties has been duly obtained in a manner satisfactory to Buyer in its sole discretion. With respect to the assignment or transfer of any Assumed Contract to Buyer that requires the consent, approval or waiver of another party seller shall use its best efforts to obtain such consent, approval or waiver of such assignment or transfer as promptly as practicable. Until such consents, approvals and waivers referred to in this Section are obtained by Seller, Seller shall (a) provide to Buyer the financial and business benefits of such Assumed Contract, and (b) enforce, for the account of Buyer, any rights of Seller arising from any such Assumed Contract (including the right to elect to terminate such Assumed Contract in accordance with the terms thereof upon the request of Buyer).

#### 2.5 Closing Consideration.

(a) Prior to the Closing Date, Buyer and Seller will agree upon a written statement (the "Estimated Closing Statement") setting forth the calculation of a good faith estimate of the Closing Date Purchase Price (the "Estimated Closing Date Purchase Price"), including all components thereof. At the Closing, Buyer will pay in cash:

- (i) the Closing Indebtedness payable to the holders of Closing Indebtedness designated in the payoff letters delivered to Buyer pursuant to Section 2.8(a)(iv);
- (ii) the Escrow Amount to the Escrow Agent, which shall be held by the Escrow Agent and released to the proper Party or Parties upon the terms and conditions set forth in the Escrow Agreement; and
- (iii) to Seller, the Closing Date Purchase Price, by wire transfer of immediately available funds, pursuant to instructions delivered to Buyer no less than three business days prior to the Closing Date.
- (b) At the Closing, Buyer Parent shall issue to Seller a common stock purchase warrant, in substantially the form attached hereto as <u>Exhibit C</u> (the "<u>Warrant</u>"), to purchase Ten Million (10,000,000) shares (the "<u>Warrant Shares</u>") of common shares, no par value per share, of Buyer Parent (the "<u>Buyer Parent Common Shares</u>"). The Warrant and the Warrant Shares are collectively referred to herein as the "<u>Buyer Securities</u>." The Warrant shall be exercisable at any time between the Closing Date and the tenth (10th) anniversary of the date of the Closing Date and shall have an exercise price equal to the closing price per share of the Buyer Parent Common Shares, as reported on the NYSE American stock exchange, either on the trading day immediate prior to the Closing Date (if the Closing occurs prior to the close of trading on the NYSE American stock exchange on the Closing Date) or on the Closing Date (if the Closing occurs after the closing of trading on the NYSE American stock exchange on the Closing Date).

#### 2.6 Determination of the Final Closing Date Purchase Price Closing Payment.

- (a) As soon as reasonably practicable, but no later than 90 days after the Closing Date, Buyer will prepare and deliver to Seller a statement (the "<u>Proposed Final Closing Statement</u>") consisting of a proposed calculation in reasonable detail of the Closing Date Purchase Price, including all components thereof (the "<u>Proposed Final Closing Date Purchase Price</u>").
- (b) If Seller disputes any amounts as shown on the Proposed Final Closing Statement, Seller shall deliver to Buyer within thirty (30) days after receipt of the Proposed Final Closing Statement a notice (the "Dispute Notice") setting forth Seller's calculation of such amounts and describing in reasonable detail the basis for the determination of such different amounts. If Seller does not deliver a Dispute Notice to Buyer within such thirty (30) day period, the Proposed Final Closing Statement prepared and delivered by Buyer shall be deemed to be the final closing statement (the "Final Closing Statement"). The Parties shall use commercially reasonable efforts to resolve such differences within a period of thirty (30) days after Seller has given the Dispute Notice. If the Parties resolve such differences, the Proposed Final Closing Statement agreed to by the Parties shall be deemed to be the Final Closing Statement. If Buyer and Seller do not reach a final resolution on the Final Closing Statement within thirty (30) days after Seller has given the Dispute Notice, unless Buyer and Seller mutually agree to continue their efforts to resolve such differences, the Neutral Accountant shall resolve such differences, pursuant to an engagement agreement among Buyer, Seller and the Neutral Accountant (which Buyer and Seller agree to execute promptly), in the manner provided below. The Neutral Accountant shall only decide the specific items under dispute by the Parties (the "Disputed Items"), solely in accordance with the terms of this Agreement. Buyer and Seller shall each be entitled to make a presentation to the Neutral Accountant, pursuant to procedures to be agreed to among Buyer, Seller and the Neutral Accountant (or, if they cannot agree on such procedures, pursuant to procedures determined by the Neutral Accountant), regarding such Party's determination of the amounts to be set forth on the Final Closing Statement; and the Parties shall use commercially reasonable efforts to cause the Neutral Accountant to resolve the differences between Buyer and Seller and determine the amounts to be set forth on the Final Closing Statement within twenty (20) days after the engagement of the Neutral Accountant. The Neutral Accountant's determination shall be based solely on such presentations of the Parties (i.e., not on an independent review) and on the definitions and other terms included herein. The Closing Statement determined by the Neutral Accountant shall be deemed to be the Final Closing Statement. Such determination by the Neutral Accountant shall be conclusive, nonappealable and binding upon the Parties, absent fraud or manifest error, and shall be considered an arbitral award for all purposes. The fees and expenses of the Neutral Accountant will be borne in the same proportion as the aggregate dollar amount of the Disputed Items that are unsuccessfully disputed by each Party (as finally determined by the Neutral Accountant) bears to the aggregate dollar amount of all of the Disputed Items submitted to the Neutral Accountant. Nothing in this Section 2.5(b) shall be construed to authorize or permit the Neutral Accountant to: (i) determine any questions or matters whatsoever under or in connection with this Agreement except for the resolution of the Disputed

Items; or (ii) resolve any such differences by making an adjustment to the Final Closing Statement that is outside of the range defined by amounts as finally proposed by Buyer and Seller.

- (c) If the Closing Date Purchase Price as finally determined in accordance with <u>Section 2.5(b)</u> (the "<u>Final Closing Date Purchase Price</u>") exceeds the Estimated Closing Date Purchase Price, then (i) Buyer shall, promptly, but no later than five (5) business days after the final determination thereof, pay such excess amount to Seller by wire transfer of immediately available funds to an account designated by Seller and (ii) Buyer and Seller shall promptly, but no later than five (5) business days after such final determination, deliver a joint written instruction to the Escrow Agent directing the Escrow Agent to release the balance of the Adjustment Escrow Amount to Seller.
- (d) If the Closing Date Purchase Price as finally determined in accordance with Section 2.5(b) is less than the Estimated Closing Date Purchase Price, then Buyer and Seller shall promptly, but no later than five (5) business days after such final determination, deliver a joint written instruction to the Escrow Agent directing the Escrow Agent to pay from the Adjustment Escrow Amount the amount of such shortfall to Buyer, and any balance of the Adjustment Escrow Amount remaining after such payment to Buyer, to Seller. If such shortfall is greater than the Adjustment Escrow Amount, then Seller shall, promptly, but no later than five (5) business days after the final determination of the Final Closing Date Purchase Price, pay the remaining balance to Buyer by wire transfer of immediately available funds to an account designated by Buyer.

(e) Any payments made pursuant to this <u>Section 2.5(b)</u> shall be treated as an adjustment to the Closing Date Purchase Price by the Parties, except as otherwise required by any Legal Requirement.

2.7 **Closing**. Upon the terms and subject to the conditions of this Agreement, the closing of the transactions contemplated hereby (the "Closing") will take place at the offices of Buyer on the third business day following the satisfaction or waiver of the conditions set forth in ARTICLE VI (other than those conditions that by their terms cannot be satisfied until the Closing, but subject to the fulfillment or waiver of such conditions at the Closing), or at such other place and time as the Parties shall mutually agree (including by means of electronic transmission of executed documents). The date of the Closing is herein called the "Closing Date." The Closing will be deemed effective as of 12:01 a.m. prevailing Eastern time on the Closing Date.

#### 2.8 Closing Deliveries.

- (a) At the Closing, Seller will deliver or cause to be delivered to Buyer the following:
  - (i) the Bill of Sale, duly executed by Seller;
  - (ii) the consents, waivers, approvals, Orders and authorizations listed on <u>Schedule 2.8(a)(ii)</u>, each in form and substance reasonably acceptable to Buyer;
  - (iii) an officer certificate, attaching (i) the Articles of Organization of Seller, certified by the Georgia Secretary of State as of a recent date prior to the Closing Date, and (ii) resolutions of Seller approved by its managers and its members authorizing the execution and delivery of, and the consummation of the transactions contemplated by, this Agreement and the Transaction Documents to which Seller is a party;
  - (iv) a certificate executed by an officer of Seller to the effect that the conditions specified in Section 6.2(a)-(d) have been satisfied;
  - (v) Payoff letters directing payment of all the Closing Indebtedness;
  - (vi) evidence satisfactory to Buyer in its sole discretion that all Liens encumbering the Acquired Assets have been or will be terminated (other than Permitted Liens);
  - (vii) the Estimated Closing Statement, duly executed by Seller;
  - (viii) the Escrow Agreement, duly executed by Seller;

- (ix) the Seller Non-competition Agreements, duly executed by Seller and the Principal Member;
- (x) Articles of Amendment to be filed with the Georgia Secretary of State on the Closing Date, amending Seller's Articles of Organization to change Seller's name to a name easily distinguishable from "Revo Squared LLC" in form and substance consistent with applicable Legal Requirements and acceptable to Buyer;

- (xi) Employment offer letters or agreements and non-solicitation and noncompetition agreements of each Key Employee and any other Employees designated by Buyer, in each case on terms and conditions satisfactory to Buyer in its sole discretion and, with respect to the Key Employees, in substantially the forms attached hereto as <u>Exhibit</u> <u>D</u> (the "Employment Agreements"), duly executed by such Key Employees and Employees;
- (xii) the IP Assignments, duly executed by Seller;
- (xiii) the form of Invention and Intellectual Property Assignment Agreement attached as <u>Schedule 2.8(a)(xiii)</u>, executed by each current and former employee and contractor of Seller or its Affiliates (the "<u>Invention Assignment</u> <u>Contracts</u>") who has contributed to the conception, reduction to practice, creation, authorship or development of any Business Intellectual Property Rights, pursuant to which such party (i) assigns to Seller its/his/her right, title and interest in and to such Business Intellectual Property Rights relating to or in the scope of such Person's employment by or engagement with Seller or its Affiliate as set forth in the Invention Assignment Contracts, and (ii) waives (subject to limitations of applicable Legal Requirements) any unassignable rights such as moral rights that he, she or it may possess in the Business Intellectual Property Rights as set forth in the Invention Assignment Contracts;
- (xiv) with respect to the Leased Real Property, an Assignment and Assumption of Lease in form and substance satisfactory to Buyer ("Assignment and Assumption of Lease") and duly executed by Seller; and
- (xv) any other certificates, documents or instruments reasonably required by Buyer to be delivered at Closing pursuant to this Agreement.
- (b) At the Closing, the Buyer Parties will deliver or cause to be delivered the following:
  - (i) the payments stated in Section 2.5(a);
  - (ii) the Estimated Closing Statement, duly executed by Buyer;
  - (iii) the Warrant;
  - (iv) the Bill of Sale, duly executed by Buyer;
  - (v) the Escrow Agreement, duly executed by Buyer;
  - (vi) the Seller Non-competition Agreements, duly executed by Buyer;
  - (vii) the Employment Agreements, duly executed by Buyer;
  - (viii) the Assignment and Assumption of Lease, duly executed by Buyer; and
  - (ix) a certificate of an officer of Buyer, dated the Closing Date, to the effect that the conditions specified in Section 6.3(a) and (b) have been satisfied.

#### 2.9 Earnout Payments.

- (a) <u>Calculation</u>. Subject to Buyer's right to off-set the Earnout Payments as set forth in this Agreement, Buyer will pay to Seller up to an aggregate of \$4,000,000 in one or more payments (the "<u>Earnout Payments</u>") based on the achievement of milestones related to future Net Sales from Revo Squared Products, as set forth below:
  - (i) If, from and after the Closing Date, Buyer's Net Sales actually received from the sale of Revo Squared Products exceed \$5,000,000 during any calendar year ending on or prior to December 31, 2027 (the period from the Closing Date to December 31, 2027 being referred to herein as the "Earnout Period"), Buyer will pay Seller a \$2,000,000 Earnout Payment after the first such occurrence only; and
  - (ii) If, from and after the Closing Date, Buyer's Net Sales actually received from the sale of Revo Squared Products exceed \$10,000,000 during any calendar year ending on or prior to December 31, 2027, Buyer will pay Seller a \$2,000,000 Earnout Payment after the first such occurrence only.

For the avoidance of doubt, each Earnout Payment will be paid only once, and therefore in no event will the Earnout Payments exceed \$4,000,000 in the aggregate. Also for the avoidance of doubt, both Earnout Payments can be earned in the same calendar year. The Earnout Payments will be paid as set forth in this Section 2.9. Following the Closing, Buyer (i) shall have complete control and sole and absolute discretion with respect to decisions concerning the operations (or ceasing of operations) of the Business, including as it relates to any or all of the Revo Squared Products, and (ii) shall be required only to take actions in connection with the Business that Buyer believes in good faith to be in the best interests of Buyer, and does not owe any duties, express or implied, to Seller by virtue of this Section 2.9 (other than to make the Earnout Payments, if any, due under this Section 2.9).

(b) Buyer shall make the initial calculation of its Net Sales from Revo Squared Products for each calendar year during the Earnout Period based upon the audited consolidated financial statements of Buyer Parent and Buyer's reasonable allocations and calculations with respect to any items not specifically set forth in such audited consolidated financial statements, and will deliver to Seller a written statement outlining Buyer's calculation of Net Sales of Revo Squared Products for a particular calendar year as soon as practicable following the end of the applicable calendar year, and in any event not later than April 15 following the end of the applicable calendar year. If Seller does not dispute such calculation within twenty (20) days after Seller's receipt of such calculation, then such calculation shall be deemed final and Buyer shall deliver payment of the Earn-Out Payments within ten (10) days after the date said calculation shall be deemed final. If Seller disputes such calculation, Seller shall deliver written notice of such dispute to Buyer within twenty (20) days after Seller's receipt of such calculation. Buyer and Seller shall provide each other with access to any information reasonably necessary to calculate the Net Sales of Revo Squared Products for the period in question. If any such dispute is not resolved within fifteen (15) days after Seller's delivery of a written notice of dispute to Buyer, Buyer and Seller shall, as promptly as practicable following the expiration of such fifteen (15) day period, jointly engage the Neutral Accountant to review the disputed item(s) with respect to such calculation. The final determination of such disputed item(s) by the Neutral Accountant shall be binding on the Parties. The cost of retaining the Neutral Accountant shall be borne equally by Buyer and Seller.

-16-

- (c) For each calendar year during the Earnout Period, Buyer agrees to keep complete and accurate records of the Net Sales from Revo Squared Products in sufficient detail to enable the Earnout Payments due hereunder to be determined. Buyer agrees to permit a representative of Seller and its agents and representatives, at Seller's expense, to periodically examine the books, ledgers, and records of Buyer during regular business hours for the purpose of and to the extent necessary to verify Seller's entitlement to the Earnout Payments.
- (d) Earn-Out Payments will be payable by Buyer as described in Section 2.9(b), above, for undisputed payments, or, for disputed payments, within twenty (20) days after resolution and determination of the Net Sales from Revo Squared Products for the applicable calendar year in which Earnout Payments are earned, as determined by Buyer and Seller or the Neutral Accountant (as applicable).
- (e) The Earnout Payments made pursuant to this Section 2.9 shall be treated as additional Purchase Price, including for Tax purposes, except as otherwise required by any Legal Requirement.

2.10 <u>Withholding</u>. Buyer, Seller or their respective Affiliates may deduct and withhold from the consideration payable in connection with the transactions contemplated by this Agreement or any other Transaction Document such amounts as Buyer, Seller or any of their respective Affiliates or agents, as applicable, is required to deduct and withhold with respect to the making of such payment under any Legal Requirement related to Taxes. Any amounts deducted and withheld in accordance with this Section 2.10 will be treated for all purposes of this Agreementand any other Transaction Document, as applicable, as having been paid to the Person in respect of which such deduction and withholding was made.

#### ARTICLE III.

# REPRESENTATIONS AND WARRANTIES OF SELLER AND PRINCIPAL MEMBER

Seller and the Principal Member hereby jointly and severally represent and warrant to the Buyer Parties as follows and, unless the context otherwise requires, each reference to "Seller" in this ARTICLE III will also refer to each of Seller's Subsidiaries, if any:

3.1 **Organization**. Seller is a corporation duly organized, validly existing and in good standing under the Legal Requirements of the State of Georgia and has all requisite corporate power and authority to own and lease the Acquired Assets and to operate and carry on the Business as presently conducted. Seller is duly qualified to do business as a foreign entity and is in good standing in each jurisdiction where the nature of the Business or the ownership of the Acquired Assets makes such qualification necessary, except where the failure to be so qualified and in good standing would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.

3.2 **Ownership**. The Principal Member owns the majority of the outstanding membership interests of Seller. Except as set forth on <u>Schedule 3.2</u>, there are no outstanding subscriptions, options, warrants, calls, contracts, demands, commitments, convertible securities or other agreements or arrangements of any character or nature under which any Seller is obligated to issue any securities of any kind representing an Equity Interest in Seller.

3.3 <u>Authority, Validity and Enforceability</u>. Seller and the Principal Member has all requisite power and authority or capacity, as applicable, to execute and deliver this Agreement and the other Transaction Documents, and to consummate the Contemplated Transactions. The execution and delivery of this Agreement and the other Transaction Documents and the consummation of the Contemplated Transactions have been duly authorized by all necessary action on the part of Seller and the Principal Member, and no other proceeding (including by their respective direct or indirect equityholders) on the part of Seller or the Principal Member is necessary to authorize Seller's or such Principal Member's execution of this Agreement or Seller's or such Principal Member's consummation of the Contemplated Transactions. This Agreement has been duly and validly executed and delivered by Seller and the Principal Member and, assuming the due authorization, execution and delivery of this Agreement by Buyer, constitutes a valid, legal and binding agreement of Seller and such Principal Member, enforceable against Seller and such Principal Member in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar Laws affecting creditors' rights and remedies generally and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity) (the "General Enforceability Exceptions")

-17-

3.4 No Subsidiaries. Seller has no Subsidiaries, and Seller is not a participant in any joint venture, partnership, or similar arrangement.

3.5 <u>No Conflict</u>. Subject to and assuming receipt of those notices, filings and approvals set forth on <u>Schedule3.6</u>, neither the execution of this Agreement or the other Transaction Documents by Seller, nor the performance by Seller of its obligations under any Transaction Document will (a) violate or conflict with any Legal Requirement applicable to such Party or by which such Party's properties or assets are bound, (b) contravene any provision contained in the Organizational Documents of Seller, (c) conflict with, violate or result in a breach (with or without the lapse of time, the giving of notice or both) of, or constitute a default (with or without the lapse of time, the giving of notice or both) or result in the creation of any Lien under (i) any Contract or (ii)" any Legal Requirement or other restriction of any Governmental Authority to which Seller is bound or to which the Seller or any of its assets or properties are subject, or (d) result in the acceleration of, or permit any Person to terminate, modify, cancel, accelerate or declare due and payable prior to its stated maturity, any material obligation of Seller.

3.6 <u>Consents</u>. No notice to, filing with, or authorization, registration, consent or approval of any Person or any Governmental Authority is necessary for the execution, delivery or performance of this Agreement or the other Transaction Documents to which Seller

or the Principal Member is a party or the consummation of the Contemplated Transactions by Seller or the Principal Member, except for notices, filings and approvals set forth on <u>Schedule3.6</u>.

# 3.7 Financial Statements; Undisclosed Liabilities.

- (a) <u>Schedule3.7</u> sets forth true and complete copies of (i) the balance sheets and statements of profit and loss of Seller as of and for the fiscal years ended December 31, 2021 and 2020 (the "<u>Annual Financial Statements</u>"), and (ii) the unaudited balance sheet and statement of profit and loss of Seller as of and for the five (5)-month period ended May 31, 2022 (the "<u>Interim Financial Statements</u>," and collectively with the Annual Financial Statements, the "<u>Financial Statements</u>"). The Financial Statements (i) have been prepared, in all material respects, in accordance with United States generally accepted accounting principles as in effect from time to time ("<u>GAAP</u>") applied on a consistent basis throughout the periods covered thereby, except for the absence of footnotes and subject to normal year-end adjustments, which adjustments shall not be material individually or in the aggregate, and (ii) fairly present, in all material respects, the consolidated financial position of Seller as of the Interim Financial Statements, the absence of footnotes and changes in cash flows for the periods then ended (subject to, in the case of the Interim Financial Statements, the absence of footnotes and normal year-end adjustments, which adjustments, which adjustments which adjustments for the advect of the advect of soft periods and changes in cash flows for the periods then ended (subject to, in the case of the Interim Financial Statements, the absence of footnotes and normal year-end adjustments, which adjustments shall not be material individually or in the aggregate). The Financial Statements have been prepared from, and are in accordance with, the books and records of Seller.
- (b) Except as set forth on <u>Schedule3.7</u>, Seller, with respect to the Business, has no Liabilities, except (i) Liabilities that are fully accrued, reserved against or reflected in the balance sheet included in the Interim Financial Statements (or disclosed in the notes thereto), all such reserves having been established consistently with past practices of the Seller, (ii) Liabilities that have arisen since the date of the balance sheet included in the Interim Financial Statements and that were incurred in the ordinary course of business consistent with past practice (the "<u>Ordinary Course of Business</u>") and which are not material in amount or (iii) Liabilities incurred in connection with this Agreement and the transactions contemplated hereby.

#### 3.8 Accounts Receivable; Account Payable.

- (a) All Accounts Receivable that are reflected on the balance sheet included in the Interim Financial Statements or on the accounting records of Seller as of the date hereof represent valid obligations arising from sales actually made or services actually performed by Seller in the Ordinary Course of Business and are current and fully collectible (calculated in a manner consistent with GAAP). There is no contest, claim, defense or right of setoff under any contract with any account debtor of an Account Receivable relating to the amount or validity of such Account Receivable. No account debtor (i) has refused or threatened in writing to refuse to pay its obligations for any reason, or (ii) to Seller's Knowledge, is insolvent or bankrupt.
- (b) All accounts payable that are reflected on the balance sheet included in the Interim Financial Statements or on the accounting records of Seller as of the date hereof represent valid obligations arising from purchases or commitments actually made by Seller in the Ordinary Course of Business and are current and payable in accordance with their terms net of the respective reserves shown on the balance sheet included in the Interim Financial Statements. There is no contest, claim, defense or right of setoff under any contract with any obligee of an account payable relating to the amount or validity of such accounts payable.

3.9 <u>Absence of Certain Developments</u>. Except as set forth on <u>Schedule3.9</u>, since December 31, 2020 through the Effective Date, (a) there has not been any Material Adverse Effect, and (b) Seller has conducted the Business in the Ordinary and Course of Business. There will not occur any action or event that, had it occurred after the Effective Date and prior to the Closing, would have required the written consent of Buyer under Section 5.2.

#### 3.10 Compliance with Laws; Governmental Authorizations; Licenses.

(a) Except as set forth on <u>Schedule3.10(a)</u>, Seller is and has been in compliance with all applicable Legal Requirements in all material respects. Except as set forth on <u>Schedule3.10(a)</u>, Seller has not received any notice regarding any actual, alleged, possible or potential violation of, or failure to comply with, any Legal Requirement applicable to the Business or the Acquired Assets.

(b) Seller has, and is in compliance in all material respects with, all Permits that are necessary to operate the Business as presently conducted, and has made available all such Permits to Buyer. All such Permits are in full force and effect and no fees or charges required to be paid with respect to such Permits are outstanding. Seller is not, and has not been since the date of issuance of the Permits, in default or violation of any such Permit.

(c) Seller, directly or indirectly through any representative or other Person acting on behalf of Seller (including any distributor, agent, or sales intermediary), has not (i) taken any action in violation of any applicable Anti-corruption Laws or (ii) paid, offered, promised to pay, or authorized the payment of, any monies or any other thing of value to any Government Official (including employees of government-owned or -controlled entities or partially governmentowned or -controlled entities), any political party or candidate for political office or to any other Person (collectively, a "Proscribed Recipient"): (A) for the purpose of (1) influencing any act, omission or decision of such Proscribed Recipient, (2) inducing such Proscribed Recipient to do or omit to do any act in violation of the lawful duty of such Proscribed Recipient, or to use his, her or its influence with a Governmental Authority to affect or influence any act or decision of such Governmental Authority, (3) assisting Seller or any of its representatives in obtaining or retaining business for or with, or directing business to, any Person in violation of any applicable Anti-corruption Laws, (4) securing any advantage in violation of any applicable Anti-corruption Laws, or (5) inducing such Proscribed Recipient to influence or affect any act or decision of any Governmental Authority in violation of any applicable Anti-corruption Laws, or (B) in a manner which would constitute or have the purpose or effect of public or commercial bribery, acceptance of, or acquiescence in extortion, kickbacks or other unlawful or improper means of obtaining or retaining business or any improper advantage. Neither Seller, the Principal Member, nor to Seller's Knowledge, none of Seller's employees or consultants has been convicted of violating any Anti-corruption Laws or, to Seller's Knowledge, subjected to any investigation or proceeding by a Governmental Authority for, in each case, potential corruption, fraud or violation of any applicable Anti-corruption Laws.

3.11 Litigation. There are no Actions or Orders pending or, to Seller's Knowledge, threatened, and since December 31, 2020, there have been no Actions or Orders against the Principal Member, Seller or an officer, director or employee of Seller (a) relating to the Business or the Acquired Assets or (b) seeking to enjoin, condition or delay the transactions contemplated under any of the Transaction Documents. None of Seller, the Principal Member or the Business is, as of the Effective Date, a party to any Action or, to Seller's Knowledge, threatened Action that would reasonably be expected to affect, prohibit, condition or delay the consummation of the Contemplated Transactions.

#### 3.12 Real Property; Person Property.

- (a) <u>Owned Real Property</u>. Seller does not own any real property nor is Seller contractually obligated to purchase any real property.
- (b) Leased Real Property. Schedule2.1(m) lists all real property or related premises currently leased or subleased by Seller, and sets forth the name of the landlord (or sublandlord, as applicable) and the street address and unit number of the Leased Real Property. True, correct and complete copies of all leases, subleases, licenses or other agreements (and all amendments, supplements or modifications to each such instrument) with respect to the Leased Real Property (collectively, the "Lease Documents") have been made available to Buyer. The Seller has a valid and enforceable leasehold interest in all Leased Real Property, in each case free and clear of all Liens, other than Permitted Liens. The Leased Real Property constitutes all real property interests of any nature whatsoever used or held for use in the operation of the Seller's business as currently conducted. The Seller is not subject to any governmental decree or Order to be sold or is being condemned, expropriated or otherwise taken by any Governmental Authority and, to the Knowledge of Seller, no such condemnation, expropriation or taking has been proposed or is contemplated. All Taxes required to be paid (including real and personal property Taxes and assessments and all special assessments, if any) pertaining to the Leased Real Property have been, and will continue to be, paid in full on or before the date that such Taxes fall due, and there are no currently existing delinquencies with respect thereto. Seller has not received any written notice of proposed local improvement charges or special levies with respect to the Leased Real Property. The Leased Real Property is supplied with utilities (including water, sewage, disposal, electricity, gas and telephone) and other services necessary for the operation of such Leased Real Property as currently operated. All buildings, structures and appurtenances comprising any part of the Leased Real Property that are currently being used in the conduct of Seller's business are in satisfactory condition and have been well maintained, normal wear and tear excepted.

(c) <u>Personal Property</u>. Seller has good and valid title or, in the case of leased assets, a valid leasehold interest or lease agreement, free and clear of all Liens, to all of the material tangible and intangible personal property and assets used by Seller in the conduct of the Business except for properties and assets disposed of in the Ordinary Course of Business since the date of the balance sheet included in the Interim Financial Statements. Seller owns or has the exclusive right to use all of the tangible personal properties and assets necessary for the operation of Seller or the conduct of the Business as conducted in the twelve (12) months prior to the date hereof and all such tangible person property is in the possession or control of Seller. All of the tangible personal property used by Seller in the conduct of the Business is (subject to ordinary wear and tear): (i) in good repair, (ii) in good operating condition and (iii) suitable for immediate use in the Ordinary Course of Business. All items of tangible personal property have been maintained in accordance with normal industry practice.

#### 3.13 Taxes.

- (a) Except as set forth on Schedule3.13(a), all Tax Returns required to be filed on or before the Effective Date by Seller, including any Tax Returns required to be filed as a result of Seller's ownership of the Acquired Assets, have been, and all Tax Returns required to be filed after the Effective Date and before the Closing Date will be, filed within the time and in the manner prescribed by applicable Legal Requirements. All such Tax Returns are true, correct and complete, and all Taxes owed or required to be paid by Seller or with respect to the Acquired Assets (whether or not shown or required to be shown on such Tax Returns) have been paid or remitted by Seller. Seller has filed Tax Returns in all jurisdictions where it is required to file Tax Returns and has not received notice of any claim by any Taxing Authority in any other jurisdiction where Seller does not file Tax Returns that Seller or any Acquired Assets are or may be subject to Taxes in that jurisdiction.
- (b) Except as set forth on Schedule 3.13, there are no Liens with respect to Taxes upon any of the Acquired Assets, other than with respect to Permitted Liens.
- (c) No audit or administrative or judicial Tax Action is pending or is currently being conducted with respect to Seller or the Acquired Assets.
- (d) Seller is not, and has not been (i) a member of any affiliated group filing or required to file a consolidated, combined, or unitary Tax Return or (ii) a party to or bound by, nor do they have or have they ever had any obligation under, any Tax sharing agreement or similar contract or arrangement. Seller does not have any Liability for the Taxes of any Person (other than Seller) under Treasury Regulation Section 1.1502-6 (or any corresponding provision of state, local or foreign Tax law), as a transferee or successor, by contract, or otherwise.
- (e) Seller has delivered to Buyer true, complete and correct copies of all Tax Returns, Tax audit reports, and statements of deficiencies for each of the last three Taxable years filed by or issued to or with respect to Seller and the Acquired Assets (or, insofar as such items relate to Seller, by or to any affiliated, consolidated, combined, or unitary group of which Seller was then a member). Schedule3.13(e) lists all of the jurisdictions (foreign and domestic) in which any Tax is properly payable by any Seller or with respect to the Acquired Assets under applicable Legal Requirements. No written rulings or agreements in respect of any Tax are pending or have been issued by or entered into with any relevant Taxing Authority with respect to Seller or any Acquired Assets.
- (f) Seller has elected to be treated as an "S corporation" (within the meaning of Sections 1361 and 1362 of the Code) for U.S. federal income tax purposes commencing in 2009, its first taxable year, and such election is effective for all Taxable periods of Seller thereafter up to and including the Closing Date. Schedule3.13(f) sets forth each state or other local jurisdiction for which Seller has made or is treated as having made an S corporation (or similar) election and its effective date, and each such election is effective for all Taxable periods of Seller thereafter up to and including the Closing Date. Neither Buyer nor Seller will be liable for any Tax under Section 1374 of the Internal Revenue Code in connection with the Contemplated Transactions.

3.14 <u>Environmental Matters</u>. At all times Seller and the Business have been in compliance with all applicable Environmental Laws and Environmental Permits, have maintained all material Environmental Permits required by applicable Environmental Laws for the operation of the Business, and have timely applied for and obtained the renewal of all such Environmental Permits. Seller has not received (i) any notice from any Governmental Authority regarding any actual or alleged violation of Environmental Laws, or any Liabilities or potential Liabilities for personal injury, property damage or investigatory or cleanup obligations arising under Environmental Laws (including in connection with the disposal of or arrangement for disposal of any Hazardous Substance), in either case which is pending or unresolved or (ii) received notice under the citizen suit provisions of any Environmental Law. There has been no release of any Hazardous Substance on any properties formerly or currently owned, leased or used by Seller that requires investigation, corrective action, cleanup or remediation by, or that has given rise to or would give rise to any Liability of, Seller under any applicable Environmental Laws. Seller has made available to Buyer all environmental assessments, and audits, and all other material environmental, health and safety documents, in the possession, custody or control of Seller or the Principal Member.

3.15 Employee Matters. Except as set forth on Schedule3.15, (i) Seller is not bound by any collective bargaining agreement with respect to any Employee, (ii) there is no labor strike, work stoppage or lockout pending or, to Seller's Knowledge, threatened against or by Seller and during the past three years there has been no such action, (iii) no union organization campaign is in progress with respect to any Employee, (iv) to Seller's Knowledge, there is no unfair labor practice charge or complaint pending or threatened against Seller before the National Labor Relations Board, and (v) there is no pending employment Action before any state or federal court, arbitrator, Equal Employment Opportunity Commission, U.S. Department of Labor or any other applicable Governmental Authority (or, to Seller's Knowledge, threat of any such Action). Seller is in compliance with all applicable Legal Requirements pertaining to employment, employment practices, employment eligibility verification and the employment of labor. Schedule 3.15 sets forth a complete and correct list of each Employee, and with respect to each such Employee, his or her (i) name; (ii) employer; (iii) title; (iv) location; (v) date of hire and years of service with Seller; (vi) exempt/non-exempt status; (vii) employment status (i.e., whether full-time, part-time, temporary, etc.); (viii) active/inactive status (and type of leave, if applicable); (ix) accrued but unused paid time off; (x) base compensation or hourly wage rate, commission/bonus and total compensation for 2020 and 2021; and (xi) current annual base salary or hourly wage rate and commission/bonus paid to date. Seller has not increased the total compensation to be paid to any Employee since January 1, 2020, except for the increases set forth on Schedule3.15. All individuals who perform services for Seller has been classified correctly, in accordance with the terms of each Employee Plan and ERISA, the Code, the Fair Labor Standards Act and all other applicable Legal Requirements, as employees (including as to overtime exempt or non-exempt), independent contractors or leased employees.

#### 3.16 Employee Benefit Plans.

- (a) Buyer shall not have any Liability after the Closing under any:
  - (i) employee benefit plan, as defined in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("<u>ERISA</u>"), maintained or contributed to by Seller or any of its Affiliates for any of its employees, former employees or directors (or their respective beneficiaries), including without limitation any group insurance or self-insured health plan, severance pay plan, non-qualified deferred compensation plan or retirement plan intended to be qualified under Code Section 401(a) (collectively, the "<u>ERISA Plans</u>");

- (ii) trust fund maintained by Seller in connection with any such ERISA Plan;
- (iii) "cafeteria plan" ("125 Plan") governed by Code Section 125; or
- (iv) other plan providing compensation, benefits or perquisites to any employees, former employees, managers, officers or directors (or their respective beneficiaries) of Seller, including without limitation any incentive, bonus, stock option, restricted stock, vacation pay or sick pay plan, in each case whether or not reduced to writing and whether funded or unfunded.
- (b) Except as set forth on <u>Schedule3.16(b)</u>, Seller maintains no group life insurance or health benefit coverage for former employees or directors, other than group life insurance or health benefit coverage mandated by Legal Requirements. Seller has timely complied with all of its "COBRA" obligations under ERISA Section 602 and Code Section 4980B; its obligations under the Uniformed Services Employment and Reemployment Rights Act of 1994, as amended; its obligations under the Family and Medical Leave Act of 1993, as amended; and its respective obligations under applicable

state insurance Legal Requirements, in each case with respect to any group life insurance and health benefit continuation coverage to be provided by those of its ERISA Plans and any 125 Plan that provide such benefits.

- (c) Each Employee Plan has been maintained, operated, and administered in compliance with its terms and any related documents or agreements and all applicable Legal Requirements. All Employee Plans are sponsored by Seller. No Employee Plan is maintained outside of the United States.
- (d) Neither Seller nor any other member of the Controlled Group currently has, and at no time in the past has had, an obligation to contribute to a "defined benefit plan" as defined in Section 3(35) of ERISA, a pension plan subject to the funding standards of Section 302 of ERISA or Section 412 of the Code, a "multiemployer plan" as defined in Section 3(37) of ERISA or Section 414(f) of the Code or a "multiple employer plan" within the meaning of Section 210(a) of ERISA or Section 413(c) of the Code or a "multiple employer welfare arrangement" as defined in Section 3(40) of ERISA.

#### 3.17 Intellectual Property Rights.

- (a) <u>Schedule 3.17(a)</u> sets forth a true, correct and complete list (with title or mark, owner, country, registration and application numbers, dates indicated and inventors, as applicable) of all Owned Intellectual Property that has been issued or registered to, or is the subject of a pending application by, the Seller or its Affiliates. All fees associated with maintaining any such Owned Intellectual Property registrations have been paid in full in a timely manner to the proper Governmental Authority. There are no materials, information, facts, or circumstances that would render any such Owned Intellectual Property registrations invalid or unenforceable, or that would materially affect any pending application for any Owned Intellectual Property registrations.
- (b) The Business Intellectual Property Rights consist of the Owned Intellectual Property and the Licensed Intellectual Property, which comprises all of the Intellectual Property Rights necessary for the Business as of the Effective Date and as of the Closing Date. Except as set forth in <u>Schedule 3.17(b)</u>, no Person who has licensed to Seller any Licensed Intellectual Property has ownership or any exclusive rights to any improvements, derivative works and other modifications made by Seller. After giving effect to the transactions contemplated by this Agreement, no Affiliate or current or former partner, director, stockholder, officer, employee or contractor of Seller or any of its Affiliates will own or retain any rights, title or interest in any Business Intellectual Property Rights. Seller is the sole and exclusive legal and beneficial owner of all right, title and interest in and to the Owned Intellectual Property, and has the valid and enforceable right to use all Licensed Intellectual Property used in or necessary for the conduct of the Business as currently conducted, in each case, free and clear of Encumbrances.

-23-

(c) The operation of the Business, as currently conducted, does not, and Buyer's continued operation of the Business, as currently conducted by Seller, will not, infringe, misappropriate, dilute or otherwise violate or conflict with any Intellectual Property Rights of any Person, otherwise violate any rights of any Person (including any right to privacy or publicity), or constitute unfair competition or deceptive trade practices or other violation of Legal Requirements. This representation and warranty does not constitute any representation or warranty with respect to any infringement, misappropriation, dilution or other violation or conflict with any Intellectual Property Rights of any Person caused by (i) any modifications to Diagnostic Software after the Closing Date, (ii) uses of the Diagnostic Software after the Closing Date that were not utilized by the Business or its customers prior to the Closing Date, and (iii) integration of the Diagnostic Software with any hardware, equipment or components that were not integrated by the Business or its customers prior to the Closing Date. Neither Seller nor any of its Affiliates have received any notice from any Person (i) alleging any of the foregoing; (ii) claiming that Seller or any of its Affiliates must license from any Person or refrain from using any Intellectual Property Rights or offering Seller or any of its Affiliates to take a license to avoid any potential claim or disruption; or (iii) challenging the validity, enforceability, patentability, registerability, or Seller's or its Affiliates' scope or ownership of any of the Business Intellectual Property Rights. To Seller's Knowledge, no Person has infringed, misappropriated, otherwise violated or conflicted with, or is suspected by Seller or any of its Affiliates to be infringing, misappropriating, otherwise violating or conflicting with, any Business Intellectual Property Rights. Prior to the Closing, Seller, and immediately after the Closing, Buyer, is not obligated to pay any royalties or similar payments to third parties with respect to the marketing, sale, distribution, manufacture, license or use of the Acquired Assets.

- (d) Except as set forth in <u>Schedule 3.17(d)</u>. Seller and its Affiliates have taken all commercially reasonable steps to protect and maintain all Business Intellectual Property Rights and to preserve the confidentiality of any trade secrets comprised in Business Intellectual Property Rights and any of their confidential information, in each case in accordance with best industry practices. All disclosures by Seller or its Affiliates of their trade secrets or confidential information have been made pursuant to a written Contract that provides reasonable protection for such confidential information and trade secrets. Neither Seller nor any Selling Stockholder shall retain any trade secret of the Business or any written or tangible description of any trade secret of the Business.
- (e) Except as set forth in Schedule 3.17(e), no current or former employee of Seller or any of its Affiliates, consultants or contractors has any valid claim of ownership, in whole or in part, to any Business Intellectual Property Rights or derivative works thereof, or has asserted any such claim of ownership or right. Except as set forth in Schedule 3.17(e), each current and former employee and contractor of Seller or its Affiliates who has contributed to the conception, reduction to practice, creation, authorship or development of any Business Intellectual Property Rights is a party to a valid, enforceable and legally binding Contract (i) assigning to Seller or its Affiliate all right, title and interest in and to such Business Intellectual Property Rights relating to or in the scope of such Person's employment by or engagement with Seller or its Affiliate, and (ii) waiving (subject to limitations of applicable Legal Requirements) any unassignable rights such as moral rights that they may possess in the Business Intellectual Property Rights.
- (f) Seller has commercially reasonable security measures and policies in place to protect confidential information, Personal Data and proprietary information, including trade secrets, software, databases, systems, networks and Internet sites, held by it from unlawful or unauthorized access, use, modification or disclosure by any Person, and Seller is in compliance with such measures and policies. To Seller's Knowledge, no Person has gained unauthorized access to or made any unauthorized use of any confidential information pertaining to Business Intellectual Property Rights, Personal Data or proprietary information maintained by Seller. At Closing, Seller shall deliver to Buyer a copy of all Source Code of all Software, and thereafter neither Seller nor any Selling Stockholder shall have any copy of the Source Code and object code of the Diagnostic Software or any portions thereof, and Selling Stockholders and Seller shall have destroyed the same. No Person other than Seller has a copy of such Source Code or object code.

-24-

- (g) Except as set forth in Schedule 3.17(g), none of the Software is subject to any "copyleft" or other obligation or condition (including any obligation or condition under any "open source" license such as the GNU Public License, Lesser GNU Public License or Mozilla Public License) that: (i) requires or could reasonably be expected to require the disclosure, licensing or distribution of any Source Code for any portion of Software; (ii) conditions or could reasonably be expected to condition the use or distribution of the Software; or (iii) otherwise imposes or could reasonably be expected to impose any material limitation, restriction or condition on the right or ability of Buyer or the Business to use or distribute any Software.
- (h) Seller has provided Buyer with true and complete copies (or in the case of any oral agreements, a complete and correct written description) of all Contracts related to Business Intellectual Property Rights, including all modifications, amendments and supplements thereto and waivers thereunder. Each such Contract is valid and binding on Seller in accordance with its terms, is in full force and effect, and is assignable without restrictions from Seller to Buyer. Neither Seller nor, to Seller's Knowledge, any other party thereto is, or is alleged to be, in breach of or default under, or has provided or received any notice of breach of, default under, or intention to terminate (including by non-renewal), any such Contract nor will any breach or default or right terminate arise by virtue of an assignment of any Business Intellectual Property Rights to Buyer.
- (i) No government funding or facilities of any educational institution or research center were used in the development of any Business Intellectual Property Rights. Seller has not obtained any U.S. government grant funding to date and is not subject to any domestic manufacturing requirement and is free to manufacture any goods in any country for the Business.

# 3.18 Material Contracts.

(a) Schedule 3.18 sets forth each Contract in effect of the type described below as of the Effective Date (all such Contracts of the type described below, the "Material Contracts"):

- (i) partnership, joint venture, or other similar contract or arrangement, or any contract relating to the acquisition or disposition of any business or assets (whether by merger, sale of stock, sale of assets, or otherwise);
- (ii) contract for the employment of any officer, individual employee or other person on a full time, part time, consulting
  or other basis (other than at-will agreements that are cancellable without severance or penalty at any time);
- (iii) agreement or indenture relating to Indebtedness or otherwise placing a Lien on any of the Acquired Assets;
- (iv) lease or agreement under which Seller is lessee of or holds or operates any property (other than real property) owned by any other party, except for any lease or agreement under which the aggregate annual rental payments do not exceed \$25,000 and the term of which does not extend beyond twelve (12) months after the Closing;

- (v) lease or agreement under which Seller is lessor of or permits any third party to hold or operate any property (other than real property) owned or controlled by such Company, except for any lease or agreement under which the aggregate annual rental payments do not exceed \$25,000 and the term of which does not extend beyond twelve (12) months after the Closing;
- (vi) contracts (other than standard purchase orders) with Seller's suppliers listed on <u>Schedule 3.21(b)</u>, if any;
- (vii) research agreements;
- (viii) contracts with third party distributors of Seller's products;
- (ix) agreement, contract or commitment concerning non-competition, non-solicitation or non-disparagement, including any contract containing covenants that in any way purport limit the ability of Seller to engage in any line of business or to compete with any Person;
- (x) collective bargaining agreement, labor contract or other written agreement, arrangement with any labor union or any employee organization;
- (xi) contract with customers pursuant to which Seller is obligated to deliver more than \$25,000 of products or services after the date hereof;
- (xii) contract with any Governmental Authority;
- (xiii) contract that involved or involves performance of services or delivery of goods or materials by Seller during the twelve months preceding or following the date of this Agreement of an amount or value in excess of \$25,000 (other than purchase orders in the Ordinary Course of Business);
- (xiv) contract that was not entered into in the Ordinary Course of Business and that involved or involves expenditures or receipts of the Companies during the twelve months preceding or following the date of this Agreement in excess of \$25,000;
- (xv) contract for the furnishing or receipt of services, the performance of which will extend over a period of more than one (1) year from the date of this Agreement;
- (xvi) contract that provides for the payment or potential payment of any severance, retention, bonus, change in control or similar compensation;
- (xvii) contract under which Seller has advanced or loaned any other Person amounts, other than advances for otherwise reimbursable business expenses in the Ordinary Course of Business;

- (xix) contract (however named) involving a sharing of profits, losses, costs or liabilities by Seller with any other Person;
- (xx) contract for capital expenditures in excess of \$25,000;
- (xxi) written guaranty and/or other similar undertaking with respect to contractual performance of third parties extended by Seller; and
- (xxii)contract entered into other than in the Ordinary Course of Business or that is material to Seller or the Business as a whole that is not otherwise required to be listed on <u>Schedule 3.18</u>.
- (b) Each Material Contract is in full force and effect and is a valid and binding agreement of Seller, and assuming such Material Contract is binding and enforceable against the other parties thereto, is enforceable in accordance with its terms (subject to the General Enforceability Exceptions). Seller is not and, to Seller's Knowledge, each of the other parties thereto are not, in default or breach under, any of such Material Contracts and no event has occurred which, with or without notice or lapse of time, or both, would constitute such a default or breach. No party to any Material Contract has delivered written notice of termination and, to Seller's Knowledge, no party has threatened to exercise any termination rights with respect to any Material Contracts. True, correct and complete copies of the Material Contracts, including all amendments, schedules, exhibits and other attachments thereto, have been made available to Buyer.

3.19 <u>Insurance</u>. <u>Schedule3.19</u> contains a complete and correct list of all policies of fire, liability, workers' compensation, property, casualty and other forms of insurance owned or held by Seller or otherwise held for the benefit of the Business as of the Effective Date. All such policies are in full force and effect. Such policies are in amounts that are customary, adequate and suitable in relation to Business, assets and Liabilities, and all premiums have been paid in full.

3.20 <u>Title to Assets</u>. Seller has good and marketable title to all of the Acquired Assets (including those reflected on the most recent balance sheet included in the Financial Statements, but excluding any such tangible assets and properties sold, consumed, or otherwise disposed of in the ordinary course of business since the date of the most recent balance sheet included in the Interim Financial Statements), free and clear of all Liens, except for Permitted Liens. The operations of the Business are conducted solely by Seller and not through any other Person. Assuming a sufficient number of Employees or suitable replacements thereof operate the Business after the Closing, the Acquired Assets are sufficient in all material respects for the continued conduct of the Business after the Closing in substantially the same manner as conducted immediately prior to the Closing and constitute all of the rights, property and assets used, or intended to be used in the Business and necessary to conduct the Business as currently conducted.

#### 3.21 Customers and Suppliers.

(a) <u>Schedule3.21(a)</u> sets forth a list of all customers of the Business since January 1, 2021 and the aggregate revenue of Seller associated with each such customer during such period. Except as set forth on <u>Schedule3.21</u>, since December 31, 2021, no such customer has delivered to Seller written notice, and to the Knowledge of Seller, there is no reason to believe that (i) any customer intends to terminate or materially reduce its relationship with Seller, (ii) any customer has disputed pricing terms or conditions (other than disputes in the ordinary course of business), (iii) any customer has disclosed a material change in the volume of goods or services to be delivered or (iv) any customer will reduce prices it pays, change its purchasing methodology or otherwise take any action or omit to take any action that would have a Materially Adverse Effect.

(b) <u>Schedule3.21(b)</u> sets forth a list of Seller's top ten (10) suppliers based on the Seller's aggregate expenditures for the calendar year ending December 31, 2021 and the aggregate expenditures associated with each such supplier during such period. Except as set forth on Schedule3.21(b), since December 31, 2021, no such supplier has delivered to Seller written notice of cancellation or termination of, or otherwise threatened in writing, and, to the Knowledge of Seller, there is no reason to believe that, that Seller supplier (A) intends to cease to supply goods or services to Seller or to otherwise terminate or materially reduce its relationship with Seller, (B) intends to discontinue, modify or substitute any services, software, goods or products (or component thereof) material to carry out the business of Seller as currently conducted, (C) intends to change or otherwise modify its pricing terms in any material respect or (D) intends to increase prices it charges or otherwise take any action or omit to take any action that would have a Materially Adverse Effect.

3.22 <u>Affiliate Transactions</u>. <u>Schedule3.22</u> sets forth (a) all existing arrangements or relationships between Seller and any of its Affiliates or representatives (other than an Employee Plan) and (b) any property or right, tangible or intangible, which is owned by an Affiliate or representative of Seller and used in the operation of the Business (collectively, "<u>Affiliate Transactions</u>").

3.23 **Brokers**. No Person is or will be entitled to a broker's, finder's, investment banker's, financial adviser's or similar fee from Seller or any of its Affiliates in connection with this Agreement or any of the Contemplated Transactions.

3.24 <u>Product Liability Claims</u>. Seller has received no claims, or incurred any uninsured or insured Liability, for or based upon breach of product warranty (other than warranty service and repair claims in the ordinary course of business not material in any amount or significance), strict Liability in tort, negligent manufacture of product, negligent provision of services or any other allegation of Liability, including or resulting in, but not limited to, product recalls, arising from the materials, design, testing, manufacturing, packaging, labeling (including instructions for use), installing, servicing or selling of Business products or from the provision of its services (hereafter collectively referred to as a "<u>Product Liability</u>"). To the Seller's Knowledge, there are no unasserted claims for, and no facts exist which could reasonably be expected to result in, a Product Liability.

3.25 <u>Warranties</u>. Seller's standard form of warranty is set forth on <u>Schedule3.25</u>. All products manufactured or sold, and all services provided by Seller prior to the Closing Date have complied, and remain in compliance, with all contractual requirements, warranties or covenants, express or implied, applicable thereto, and complied and remain in compliance with all applicable governmental or regulatory specifications therefor and applicable thereto at the time of manufacture or sale. Each of the products produced or sold or services provided by Seller (in the case of products, when produced and sold, and in the case of services, when provided): (i) complied with all then applicable Legal Requirements; and (ii) conformed to any written express or implied warranties, and will continue to comply until the expiration date of such warranties. Seller has not provided any oral warranties, promises or representations with respect to any Business product or service.

#### 3.26 Securities Law.

(a) Investment Intent; Restrictions on Transfer. Seller understands that the Warrants are "restricted securities" and have not been registered under the Securities Act or any applicable state securities law and is acquiring the Warrants and, upon exercise of the Warrants, will acquire the Warrant Shares issuable upon exercise thereof, as principal for its own account and not with a view to, or for distributing or reselling such Buyer Securities or any part thereof in violation of the Securities Act or any applicable state securities laws. Seller does not presently have any agreement, plan or understanding, directly or indirectly, with any Person to distribute or effect any distribution of any of the Buyer Securities (or any securities which are derivatives thereof) to or through any person or entity. Seller agrees not to sell, assign, transfer or otherwise dispose of the Buyer Securities unless a registration statement relating thereto has been duly filed and become effective under the Securities Act, and applicable state or foreign securities laws, or unless in the opinion of counsel satisfactory to Buyer Parent no such registration is required under the circumstances. Seller acknowledges that the Warrants, and certificates evidencing the Warrant Shares shall bear any legend as required by the "blue sky" laws of any applicable state and a customary restricted stock legend.

<sup>(</sup>b) <u>Seller Status</u>. At the time that Seller was offered the Buyer Securities, Seller was, and at the date hereof Seller is an "accredited investor" as defined in Rule 501(a) under the Securities Act.

- (c) <u>General Solicitation</u>. Seller is not purchasing the Buyer Securities as a result of any advertisement, article, notice or other communication regarding the Buyer Securities published in any newspaper, magazine or similar media or broadcast over television or radio or presented at any seminar or any other general advertisement or general solicitation.
- (d) **Experience of Seller**. Seller, either alone or together with its representatives, has such knowledge, sophistication and experience in business and financial matters so as to be capable of evaluating the merits and risks of the prospective investment in the Buyer Securities, and has so evaluated the merits and risks of such investment. Seller is able to bear the economic risk of an investment in the Buyer Securities and, at the present time, is able to afford a complete loss of such investment.
- (e) <u>Access to Information</u>. Each Seller Party acknowledges that such Seller Party has had the opportunity to review the reports, schedules, forms, statements and other documents required to be filed by it under the Exchange Act, including pursuant to Section 13(a) or 15(d) thereof, and has been afforded (i) the opportunity to ask such questions as Seller Party has deemed necessary of, and to receive answers from, representatives of the Buyer Parties concerning the Buyer Parties and the merits and risks of investing in the Buyer Securities; (ii) access to information about the Buyer Parent and its financial condition, results of operations, business, properties, management and prospects sufficient to enable it to evaluate its investment; and (iii) the opportunity to obtain such additional information that Buyer Parent possesses or can acquire without unreasonable effort or expense that is necessary to make an informed investment decision with respect to the investment.
- (f) **No Governmental Review**. Each Seller Party understands that no United States federal or state agency or any other government or governmental agency has passed on or made any recommendation or endorsement of the Buyer Securities or the fairness or suitability of the investment in the Buyer Securities nor have such authorities passed upon or endorsed the merits of the offering of the Securities.
- (g) <u>Canadian Securities Matters</u>. Seller represents and warrants that (i) Seller is not a resident of Canada or a person or entity in Canada; (ii) the Warrant and the Warrant Shares, when issued, are being issued and sold pursuant to exemptions from the prospectus and registration requirements under applicable Canadian securities laws; (iii) the Warrant is not being, and the Warrant Shares will not be, acquired with a view to the distribution of any of these securities to a resident of Canada or a person or entity in Canada; and (iv) no resale of the Warrant or the Warrant Shares during the four-month "restricted period" following the date on which the Warrant is issued will be made to a resident of Canada or a person or entity in Canada.

3.27 **Disclosure**. All disclosures provided to Buyer regarding Seller, the Business and the transactions contemplated hereby, including the Disclosure Schedule, furnished by or on behalf of Seller with respect to the representations and warranties made herein are true and correct with respect to such representations and warranties and do not contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading.

3.28 <u>No Other Representations</u>. Except for the representations and warranties contained in this ARTICLE III, including the related portions of the Disclosure Schedule, neither the Company nor the Principal Member has made or makes any other express or implied representation or warranty, written or oral, on behalf of Seller, the Acquired Assets, or the Business.

-29-

#### ARTICLE IV. REPRESENTATIONS AND WARRANTIES OF THE BUYER PARTIES

The Buyer Parties represents and warrants to Seller as follows:

4.1 **Organization**. Buyer is a corporation duly incorporated, validly existing and in good standing under the Legal Requirements of the State of Delaware and has all requisite power and authority to own, lease and operate its property and assets and to carry on its business as presently conducted.

4.2 <u>Authority, Validity and Enforceability</u>. Buyer has all requisite corporate power and authority or capacity, as applicable, to execute, deliver and perform its respective obligations under this Agreement and the other Transaction Documents to which it is a

party. The execution, delivery and performance of this Agreement and each of the Transaction Documents, and the consummation of the transactions contemplated hereby, have been duly authorized and approved by all required action of Buyer or Buyer Parent, as applicable. This Agreement and each of the other Transaction Documents have been duly executed and delivered by Buyer or Buyer Parent, as applicable, and, assuming due authorization, execution, delivery and performance by Seller and the Principal Member, represent the legal, valid and binding obligation of Buyer enforceable against such Parties in accordance with its terms, subject to the General Enforceability Exceptions. No further action on the part of Buyer is or will be required in connection with the transactions contemplated by this Agreement or the other Transaction Documents, except as specified in the Warrant.

4.3 <u>No Conflict</u>. Neither the execution and delivery of this Agreement or any other Transaction Document, the consummation of the transactions contemplated under any Transaction Document, nor the fulfillment of and the performance by the Buyer Parties of their respective obligations hereunder or thereunder will (a) contravene any provision contained in the Buyer Parties' Organizational Documents, (b) conflict with, violate or result in a breach (with or without the lapse of time, the giving of notice or both) of, or constitute a default (with or without the lapse of time, the giving of notice or both) or result in the creation of any Lien under (i) any material Contract or (ii) assuming satisfaction of the requirements set forth in Section 4.4, any Legal Requirement or other restriction of any Governmental Authority to which either Buyer Party is bound or to which the Buyer Parties' assets or properties are subject or (c) result in the acceleration of, or permit any Person to terminate, modify, cancel, accelerate or declare due and payable prior to its stated maturity, any material obligation of either Buyer Party, which, in the case of clauses (b) and (c) would not, individually or in the aggregate, reasonably be expected to have a material adverse effect on the ability of such Buyer Party to perform its obligations under this Agreement or to consummate the transactions contemplated by the Transaction Documents.

4.4 **Consents**. Except for a Supplemental Listing Application required to be filed with NYSE American stock exchange related to the Warrant Shares, no notice to, filing with, or authorization, registration, consent or approval of any Governmental Authority is necessary for the execution, delivery or performance of this Agreement or the consummation of the transactions contemplated by the Transaction Documents by the Buyer Parties, except for those the failure of which to make or obtain would not individually or in the aggregate, reasonably be expected to have a material adverse effect on the ability of the Buyer Parties to perform their respective obligations under this Agreement or to consummate the Contemplated Transactions.

4.5 **Issuance of the Securities**. The Warrants have been duly authorized and, when issued in accordance with the terms of the Transaction Documents, will be duly and validly issued, free and clear of all Liens, other than restrictions on transfer provided for in the Transaction Documents or imposed by applicable securities laws, and shall not be subject to preemptive or similar rights of shareholders. The Warrant Shares issuable upon exercise of the Warrants have been duly authorized and, when issued and paid for in accordance with the terms of the Transaction Documents and the Warrants, will be duly and validly issued, fully paid and nonassessable, free and clear of all Liens, other than restrictions on transfer provided for in the Transaction Documents or imposed by applicable securities laws, and shall not be subject to preemptive or similar rights of shareholders. Assuming the accuracy of the representations and warranties of Seller in this Agreement, the Warrant and the Warrant Shares will be issued in compliance with applicable federal and state securities laws and Canadian securities laws.

4.6 **Brokers**. No Person is or will be entitled to a broker's, finder's, investment banker's, financial adviser's or similar fee from Buyer in connection with this Agreement or any of the transactions contemplated hereby.

4.7 <u>Solvency</u>. Immediately after giving effect to the Contemplated Transactions, Buyer shall be solvent and shall: (a) be able to pay its debts as they become due; (b) own property that has a fair saleable value greater than the amounts required to pay its debts (including a reasonable estimate of the amount of all contingent liabilities); and (c) have adequate capital to carry on its business. Buyer is paying the cash portion of the Closing Date Purchase Price out of its available cash and is not incurring Indebtedness in order to fund Closing Date Purchase Price.

4.8 **Independent Investigation**. Buyer acknowledges and agrees in making its decision to enter into this Agreement and to consummate the transactions contemplated hereby, Buyer has relied solely upon its own investigation and the express representations and warranties of the Seller set forth in Article III of this Agreement (including the related portions of the Disclosure Schedules) and no other express or implied representations, warranties or omissions.

4.9 <u>No Other Representations</u>. Except for the representations and warranties contained in this ARTICLE IV, neither Buyer Party has made or makes any other express or implied representation or warranty, written or oral.

#### ARTICLE V. COVENANTS AND AGREEMENTS

#### 5.1 Access and Information.

- (a) From the Effective Date until the Closing or, if earlier, the termination of this Agreement in accordance with its terms, Buyer will be entitled to access the properties, business and operations of Seller and the Business (including access to the personnel, counsel, accountants, consultants and representatives thereof) and such examination of the books and records of Seller and the Business as Buyer may reasonably request, during normal business hours and upon reasonable advance notice, and, with Seller's prior permission, Buyer may personally contact Employees, customers, suppliers and representatives selected by Buyer. Seller will cooperate with any such requests for access to the foregoing to the extent such access does not unreasonably interfere with the operations, activities and Employees of Seller.
- (b) All information disclosed, whether before or after the Closing, pursuant to this Agreement or the other Transaction Documents or in connection with the transactions contemplated by any Transaction Document will be kept confidential by such Persons in accordance with and subject to the Confidentiality Agreement and will not be used by any Person, other than in connection with the transactions contemplated by this Agreement. The Parties agree that notwithstanding anything to the contrary contained in the Confidentiality Agreement, the Confidentiality Agreement will survive from the Effective Date, and only if the Closing occurs, the confidentiality obligations with respect to any Acquired Assets that constitute "confidential information" thereunder will terminate at the Closing.

5.2 <u>Conduct of Business Prior to Closing</u>. From the Effective Date until the Closing or, if earlier, the termination of this Agreement in accordance with its terms, except as (w) otherwise expressly provided herein, (x) expressly consented to in writing by Buyer, (y) set forth on <u>Schedule5.2</u> or (z) expressly required by applicable Legal Requirement, Seller and the Principal Member will, and will cause Seller's affiliates, members, director's, officers, employees, agents and representative to:

- (a) (i) use its commercially reasonable efforts to preserve the present business operations, organization and goodwill of the Business and preserve present relationships with customers, suppliers and employees of the Business and (ii) conduct the Business in the Ordinary Course of Business;
- (b) not take or permit any action that, if it had been taken or permitted prior to the Effective Date, would reasonably be expected to result in a breach of any representation or warranty made by Seller or the Principal Member in this Agreement;
- (c) not amend its Organizational Documents in any manner that would frustrate the transactions contemplated by this Agreement;
- (d) not merge or consolidate with or otherwise acquire the Equity Interests of any other Person;
- (e) not adopt a plan or complete or partial liquidation or authorize or undertake a dissolution, consolidation, restructuring, recapitalization or other reorganization;
- (f) not sell, pledge, dispose of, transfer, lease, license, guarantee, encumber or authorize the sale, pledge, disposition, transfer, lease, license, guarantee or encumbrance of any assets, other than any sale of inventory in the Ordinary Course of the Business;
- (g) not pledge, dispose of, transfer, lease, license, guarantee, encumber or authorize the sale, pledge, disposition, transfer, lease, license, guarantee or encumber any Equity Interests in Seller, or issue any subscriptions, options, warrants, calls, contracts, demands, commitments, or convertible securities under which any Seller is obligated to issue any Equity Interests of Seller.

- (h) not take any action that would reasonably be expected to materially increase Taxes with respect to the Business or the Acquired Assets for any taxable period beginning after the Closing Date or the portion of any Straddle Period beginning the day after the Closing Date;
- (i) not waive, release, compromise or settle any pending or threatened Action except for Actions with respect to which an insurer has the sole right to control the decision to settle; or
- (j) agree, commit or offer to or fail to perform any action that results in or legally binds Seller to do any of the foregoing referred to in clauses (a)-(i) of this Section 5.2.

#### 5.3 Commercially Reasonable Efforts; Notification of Certain Events.

- (a) Subject to the terms and conditions herein provided, each of the Parties will use its commercially reasonable efforts to take, or cause to be taken, all action, and to do, or cause to be done, all things reasonably necessary, proper or advisable under applicable Legal Requirements to consummate and make effective the Contemplated Transactions. Each of the Parties will use its commercially reasonable efforts to promptly obtain all authorizations, consents, orders and approvals of, and to promptly give all notices to and make all filings with, all Governmental Authorities and other Persons that may be or become necessary for the performance of its obligations under this Agreement and the consummation of the Contemplated Transactions, including the consents, waivers, approvals and notices referred to in <u>Schedule3.6</u>. All such consents, waivers, approvals and notices shall be in writing and in form and substance reasonably satisfactory to Buyer. Executed counterparts of such consents, waivers and approvals shall be delivered to Buyer promptly after receipt thereof, and copies of such notices shall be delivered to Buyer promptly after the making thereof. Notwithstanding the foregoing or anything to the contrary set forth in this Agreement, in connection with obtaining such authorizations, consents, orders and approvals from third parties that are not Governmental Authorities, no Party will be required to make payments to any Person, provided, that Seller will remain required to make payments required by the terms of any Contract between Seller and such Person.
- (b) Each Party will give prompt written notice to the other of the occurrence, or failure to occur, of any event which occurrence or failure would cause any representation or warranty of Seller or Buyer as the case may be, contained in the Transaction Documents to be untrue or inaccurate in any material respect at any time from the Effective Date to the Closing or that would reasonably be expected to result in the failure to satisfy any of the conditions specified in ARTICLE VI, or otherwise prohibit, condition or delay the Closing. Notwithstanding anything herein to the contrary, in no event will any notice provided under this Section 5.3(b) be deemed to cure any breach of any such representation or warranty for any purpose under this Agreement.

5.4 **Public Announcements**. The timing and content of all public announcements regarding any aspect of this Agreement or the transactions contemplated hereby will be agreed upon in advance by the Parties; provided that any Party may make any such announcement or disclosure required by Legal Requirement or securities listing standard (in which case each Party will consult with the other Parties prior to any such announcement or disclosure to the extent reasonably practicable as to the form and content of such disclosure or announcement and will only disclose that information that is required by Legal Requirement based upon advice of counsel).

#### 5.5 Employee Matters.

- (a) Buyer may offer employment from and after the Effective Date to any Employee as Buyer may determine in its sole discretion, on such terms and conditions as Buyer may determine in its sole discretion, but Buyer will not be obligated to do so pursuant to this Agreement or for any other reason.
- (b) Seller will terminate the employment of all Transferred Employees, effective immediately prior to the Closing. Buyer shall not and does not assume any of Seller's employment obligations, wage or salary payment obligations, including without limitation those arising under any pension, profit sharing, deferred compensation, severance, welfare, sick leave, accrued or earned vacation, wage or other employee benefit plan, procedure, policy or practice of Seller, and those arising under any Contract to pay any Employee any proceeds from the transactions contemplated hereby ("Employment Obligations") regardless of whether such plan, procedure, policy or practice is disclosed in this Agreement or the Disclosure Schedule. Seller shall pay or otherwise satisfy, at or prior to the Closing Date, all accrued and unpaid (or

<sup>-32-</sup>

unsatisfied) Employment Obligations as of such time, regardless of whether the Employment Obligations are disclosed by Seller or otherwise mentioned in this Agreement.

- (c) (i) Seller will have the sole responsibility for "continuation coverage" benefits provided on and after the Closing Date for all current and former employees of Seller or any other member of the Controlled Group and "qualified beneficiaries" of such employees for whom a "qualifying event" occurs prior to or on the Closing Date, and (ii) Seller will have the sole responsibility for "continuation coverage" benefits provided after the Closing Date for all current and former employees of Seller or any other member of the Controlled Group who do not become Transferred Employees and "qualified beneficiaries" of such employees for whom a "qualifying event" occurs after Closing Date. Buyer will have the sole responsibility for "continuation coverage" benefits provided after the Closing Date for all Transferred Employees and "qualified beneficiaries" of such employees for whom a "qualifying event" occurs after Closing Date. Buyer will have the sole responsibility for "continuation coverage" benefits provided after the Closing Date for all Transferred Employees and "qualified beneficiaries" of Transferred Employees for whom a "qualifying event" occurs after the Closing Date. The terms "continuation coverage," "qualified beneficiaries" and "qualifying event" shall have the meanings ascribed to them under section 4980B of the Code and sections 601-608 of ERISA and any similar state Legal Requirement.
- (d) Following the Effective Date, Seller will reasonably cooperate in all matters reasonably necessary to effect the transactions contemplated by this Section 5.5, including, to the extent permitted by applicable Legal Requirement, exchanging information and data relating to payroll, workers compensation, employee benefits and employee benefit plan coverages, and in obtaining any governmental approvals required hereunder.
- (e) This Agreement is not intended by the Parties to, and nothing in this Section 5.5 or otherwise in this Agreement, whether express or implied, will (i) constitute an amendment to any Employee Plan or any employee benefit plan of Buyer or its Affiliates or (ii) confer on any current or former employee of Seller or any other Person (other than the parties to this Agreement) any rights or remedies (including third-party beneficiary rights).

5.6 Exclusivity. Seller shall, and shall cause its representatives and Employees to, immediately cease and cause to be terminated all existing activities, discussions or negotiations with any Persons (other than the Buyer Parties) with respect to, or that would reasonably be expected to lead to, any Acquisition Proposal. From and after the Effective Date until the earlier of Closing or the termination of this Agreement in accordance with ARTICLE VII, the Principal Member shall not, and Seller shall not and shall direct its employees and representatives not to, directly or indirectly, (i) initiate, facilitate, solicit or encourage (including by way of furnishing non-public information), directly or indirectly, inquiries or proposals that constitute, or could reasonably be expected to lead to, any Acquisition Proposal, (ii) initiate, engage or participate in any way with any third party in any discussions or negotiations regarding, or furnish or disclose any non-public information to any third party in connection with, or take any other action to knowingly facilitate any inquiries or the making of any proposal that constitutes, or could reasonably be expected to lead to, any Acquisition Proposal (except to notify such third party of the existence of the provisions of this Section 5.6), or (iii) enter into any agreement with respect to any Acquisition Proposal or approve or resolve to approve any Acquisition Proposal, or enter into any agreement, arrangement or understanding that would require the Principal Member or Seller to abandon, terminate or fail to consummate the transactions contemplated by this Agreement. Without limiting the foregoing, it is agreed that any violation of the foregoing restrictions by any employee or representative, whether or not such Person is purporting to act on behalf of Seller, or otherwise, will be deemed to be a breach of this Section 5.6 by Seller, and Seller will cause its employees and representatives to comply with the terms of this Section 5.6. From and after the Effective Date until the earlier of Closing or the termination of this Agreement, the Principal Member and Seller shall promptly (and in any event within one calendar day following and three business days prior to providing any such Person with any information) notify Buyer in the event that Seller or the Principal Member receives, directly or indirectly: (i) any Acquisition Proposal; (ii) any request for non-public information relating to Seller by any Person that informs Seller or its Representatives that such Person is considering making, or has made, an Acquisition Proposal; or (iii) any request for discussions or negotiations relating to a possible Acquisition Proposal. Such notice shall be made orally and confirmed in writing, and shall indicate the material terms and conditions thereof and the identity of the other party or parties involved.

5.7 Tax Matters.

- (a) Seller shall bear all Transfer Taxes incurred, imposed, assessed, or payable against any Party in connection with or as a result of this Agreement. Each Party shall cooperate as reasonably requested to minimize such Transfer Taxes, including with respect to any available exemption from such Transfer Taxes. Seller shall prepare and file all applicable Transfer Tax returns.
- (b) Each Party will cooperate fully, as and to the extent reasonably requested by the other Party, in connection with the preparation, execution and filing of Tax Returns pursuant to this Section 5.7 and any Actions with respect to Taxes. Such cooperation will include the retention and (upon the other Party's request) the provision of records and information that are reasonably relevant to any such Tax Returns and Actions and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder; provided, however, that in no event will Buyer be required to provide a copy of any Tax Returns of Buyer or its Affiliates to any other Person. Seller agrees (i) to retain all books and records with respect to Tax matters pertinent to the Acquired Assets or the Business relating to any Taxable period beginning before the Closing Date until three months following the expiration of the statute of limitations (and any extensions thereof) of the respective Taxable periods, and to abide by all record retention agreements entered into with any Taxing Authority, and (ii) to give Buyer reasonable written notice prior to transferring, destroying or discarding any such books and records and, if Buyer so requests, Seller will allow Buyer to take possession of such books and records.
- (c) Prior to the Closing, Buyer and Seller shall agree on the method of allocation of the Purchase Price (and other applicable amounts for Tax purposes) among the Acquired Assets, in accordance with Section 1060 of the Code and the Treasury Regulations thereunder (and any similar U.S. state or local Legal Requirement), in the manner to be set forth on <u>Schedule5.7(c)</u> and to be attached to this Agreement prior to or at the Closing (the "<u>Allocation</u>"). Each Party will prepare and file, and cause such Party's Affiliates to prepare and file, all Tax Returns on a basis consistent with the Allocation (including IRS Form 8594 and any equivalent U.S. state or local forms) and will take no position, and cause such Party's Affiliates to take no position, inconsistent with the Allocation on any Tax Return or in any audit or other Action with respect to Taxes or otherwise. If the Allocation is disputed by any Taxing Authority, the Party receiving notice of the dispute will promptly notify the other Parties of such dispute, and each Party will use its commercially reasonable efforts to defend such Allocation in any audit or other Action and not to settle or otherwise dispose of such audit or other Action without the prior written consent of the other Parties (such consent not to be unreasonably withheld, conditioned or delayed).
- (d) The Parties hereby waive compliance with the provisions of any bulk sales, bulk transfer or similar Laws of any jurisdiction that may otherwise be applicable with respect to the sale of any or all of the Acquired Assets to Buyer; it being understood that any Liabilities arising out of the failure of Seller to comply with the requirements and provisions of any bulk sales, bulk transfer or similar Laws of any jurisdiction which would not otherwise constitute Assumed Liabilities shall be treated as Retained Liabilities.
- (e) Notwithstanding anything to the contrary in this Agreement, each Party will reimburse the other Parties for any Taxes paid by such Party (or any of its Affiliates) that are the responsibility of the other Parties pursuant to this Agreement within ten business days after payment of such Taxes.

-35-

# 5.8 Wrong Pockets; Transfer of Certain Funds Received Post-Closing.

- (a) If either Buyer or Seller becomes aware that any of the Acquired Assets has not been transferred to Buyer or that any of the Excluded Assets has been transferred to Buyer, it shall promptly notify the other and the Parties shall, as soon as reasonably practicable, ensure that such property is transferred, at the expense of the Party that is seeking the assets to be transferred to it and with any necessary prior third-party consent or approval, to (a) Buyer, in the case of any Acquired Asset which was not transferred to Buyer at the Closing; or (b) Seller, in the case of any Excluded Asset which was transferred to Buyer at the Closing.
- (b) With respect to any and all amounts received or collected by Seller from and after the Closing (a) attributable to, or in respect of, any Acquired Asset, including the Accounts Receivable, or the Business (other than the Excluded Assets) and (b) which become the property of Buyer as a result of the consummation of the transactions contemplated by this Agreement, Seller shall provide notice of such receipt or collection to Buyer and pay promptly (and in any event within

five business days of their receipt or collection) to Buyer any and all such amounts so received or collected by wire transfer of immediately available funds to an account specified by Buyer or by other means acceptable to Buyer.

## 5.9 Post-Closing Transition Services.

- (a) For a period of up to nine (9) months following the Closing Date, if requested by Buyer, Seller shall support transfer of manufacturing of Revo Squared Products by (i) introducing Buyer to representatives of all suppliers utilized by Seller in the manufacturing of the Revo Squared Products, including without limitation suppliers of digital x-ray and ultrasound products, and (ii) using Seller's best efforts to cause each such supplier to supply goods to Buyer (to the extent it is not already doing so), including without limitation, waiving any confidentiality, non-competition or other restrictions in Seller's supplier agreements that would prohibit such suppliers to serve as a supplier to Buyer after the Closing.
- (b) For a period of up to eighteen (18) months following the Closing Date, if requested by Buyer, Seller shall assist Buyer in manufacturing Revo Squared Products by procuring Revo Squared Products and components used in the manufacturing thereof for Buyer's benefit and at Buyer's expense.

5.10 **Further Assurances**. Subject to the terms and conditions of this Agreement, upon the request of Buyer or Seller, each of the Parties will do, execute, acknowledge and deliver all such further acts, assurances, deeds, assignments, transfers, conveyances and other instruments and papers as may be reasonably required or appropriate to carry out the transactions contemplated by this Agreement.

-36-

# ARTICLE VI. CONDITIONS TO CLOSING

6.1 <u>Mutual Conditions</u>. The respective obligations of each Party to consummate, or cause to be consummated, the transactions contemplated by this Agreement, is subject to, at or prior to the Closing, there being no: (i) Legal Requirement of any nature issued by a Governmental Authority of competent jurisdiction that restrains, enjoins or otherwise prohibits, or has the effect of restraining, enjoining or otherwise prohibiting, the transactions contemplated by any Transaction Document from being consummated as herein provided, or (ii) pending Action seeking to prohibit, prevent, make illegal, delay or otherwise interfere with the consummation of any of the transactions contemplated by this Agreement.

6.2 <u>Conditions to the Obligations of the Buyer Parties</u>. The obligations of the Buyer Parties to consummate the transactions contemplated by this Agreement will be subject to the fulfillment prior to or at Closing of each of the following conditions, any and all of which may be waived, in whole or in part, by the Buyer Parties to the extent permitted by applicable Legal Requirement:

- (a) <u>Representations and Warranties</u>. (i) The Fundamental Representations of Seller and the Principal Member shall be true and correct as of the Effective Date and as of the Closing Date, and (ii) all other representations and warranties set forth in ARTICLE III shall be true and correct in all material respects (without regard to any qualifications therein as to "materiality" or Material Adverse Effect) as of the Effective Date and as of the Closing Date (or as of the specific date referred to for any representation or warranty which specifically refers to an earlier date).
- (b) <u>Covenants</u>. Seller and the Principal Member shall have duly performed or complied with, in all material respects, all of the covenants, obligations and conditions to be performed or complied with by such Parties under the terms of this Agreement prior to or at Closing.
- (c) <u>Consents and Approvals</u>. All governmental and regulatory approvals and consents of contracting parties, requisite or appropriate to the consummation of the Contemplated Transactions shall have been obtained, and such consents or approvals shall remain in full force and effect.
- (d) <u>Material Adverse Effect</u>. Since the Effective Date, no Material Adverse Effect shall have occurred.
- (e) <u>Receipt of Permits</u>. Buyer shall have received all Permits necessary to operate the Business after the Closing Date as presently conducted.

- (f) <u>Transaction Documents</u>. Seller and the Principal Member shall have delivered to Buyer duly executed counterparts to the Transaction Documents and all other documents and deliveries as are set forth in Section 2.8(a).
- (g) <u>Employment Agreements</u>. Each of the Employees required by Buyer shall have agreed to become a Transferred Employee and entered into an Employment Agreement with Buyer or one of its Affiliates, in each case on terms and conditions satisfactory to Buyer in its sole discretion.
- (h) <u>Due Diligence Board Approval</u>. Buyer shall be satisfied in its sole discretion with the results of its due diligence regarding the Business and Buyer shall have received approval from its board of directors to enter into this Agreement and consummate the Contemplated Transactions.

6.3 <u>Conditions to the Obligations of Seller Parties</u>. The obligations of the Seller Parties to consummate the transactions contemplated by this Agreement will be subject to the fulfillment at or prior to the Closing of each of the following conditions, any and all of which may be waived in whole or in part by the Seller Parties to the extent permitted by applicable Legal Requirement:

- (a) <u>Representations and Warranties</u>. (i) The Fundamental Representations of Buyer shall be true and correct as of the Effective Date and as of the Closing Date and (ii) all other representations and warranties set forth in ARTICLE IV shall be true and correct in all material respects (without regard to any qualifications therein as to "materiality") as of the Effective Date and as of the Closing Date (or as of the specific date referred to for any representation or warranty which specifically refers to an earlier date).
- (b) <u>Covenants</u>. The Buyer Parties shall have performed, in all material respects, all of the covenants and obligations which are to be performed by it under the terms of this Agreement at or prior to the Closing.
- (c) <u>Transaction Documents</u>. The Buyer Parties shall have delivered to Seller duly executed counterparts to the Transaction Documents and all other documents and deliveries as are set forth in Section 2.8(b).
- (d) <u>Employment Agreements</u>. Each Key Employee shall have entered into an Employment Agreement with Buyer or one of its Affiliates, in each case on terms and conditions satisfactory to such Key Employees.

### -37-

### **ARTICLE VII.** TERMINATION

7.1 <u>Termination</u>. This Agreement may be terminated and the transactions contemplated herein may be abandoned at any time prior to Closing:

- (a) by mutual written consent of the Buyer Parties and Seller;
- (b) by either Buyer or Seller, if the Closing will not have been consummated on or before the date that is 45 days after the Effective Date (as may be extended pursuant to the terms hereof, the "<u>Termination Date</u>"), unless extended by written agreement of such Parties; provided, however, that the right to terminate this Agreement under this paragraph will not be available to any Party whose failure to fulfill or comply with any obligation or covenant under this Agreement has been the cause of, or resulted in, the failure of the Closing to occur on or prior to such date;
- (c) by Buyer (so long as Buyer is not then in material breach of any of its representations, warranties or covenants contained in this Agreement), if a breach of this Agreement by Seller or the Principal Member results or would result in any of the conditions set forth in Section 6.1 or Section 6.2 not being satisfied and such breach cannot be cured or, if curable, remains uncured for a period of 30 days after Seller has received written notice from Buyer of the occurrence of such breach (or, if earlier, the Termination Date), and such conditions have not been waived by Buyer;
- (d) by Seller or the Principal Member (so long as Seller or the Principal Member are not then in material breach of any of their representations, warranties or covenants contained in this Agreement), if a breach of this Agreement by the Buyer Parties results or would result in any of the conditions set forth in Section 6.1 or Section 6.3 not being satisfied and such breach cannot be cured or, if curable, remains uncured for a period of 30 days after Buyer has received written notice

from Seller of the occurrence of such breach (or, if earlier, the Termination Date), and such conditions have not been waived by Seller or the Principal Member;

(e) by any Party, if any Governmental Authority will have issued an Order or Legal Requirement enjoining or otherwise prohibiting the transactions contemplated by this Agreement and such Order or Legal Requirement will have become final and nonappealable, except that the right to terminate this Agreement and abandon the transactions contemplated by this Agreement under this paragraph will not be available to any Party whose failure to fulfill or comply with any obligation or covenant under this Agreement has been the cause of, or resulted in, the issuance of such nonappealable Order or Legal Requirement;

7.2 Effect of Termination. If this Agreement is terminated pursuant to Section 7.1 hereof, (a) all rights and obligations of the Parties hereunder will terminate and no Party will have any liability to the other, except for obligations of the Parties in Section 5.1, this Section 7.2, and ARTICLE IX, which will survive the termination of this Agreement and (b) termination will not relieve any Party from liability for any intentional or willful breaches of this Agreement prior to the date of such termination. An "intentional or willful breach" means a breach or failure to perform, in each case, that is the consequence of an act or omission by a Party with the intention that the taking of such act or failure to take such act would, or would cause a breach of this Agreement.

-38-

## ARTICLE VIII. SURVIVAL; INDEMNIFICATION

8.1 Survival of Representations, Warranties and Covenants. Each of the representations and warranties set forth in this Agreement will survive the Closing and remain in full force and effect for a period of fifteen (15) months after the Closing Date, except that the Fundamental Representations will survive the Closing and remain in full force and effect until the earlier of (i) the expiration of the applicable statute of limitations or (ii) six (6) years following the Closing Date. The date upon which any representation or warranty contained herein terminates if any, is referred to herein as the "Survival Date" for such representation or warranty. None of the covenants or other agreements contained in this Agreement shall survive the Closing other than those which by their express terms are to be performed after Closing, and each such surviving covenant and other agreement of each Party set forth herein shall survive (i) in accordance with its terms or (ii) if such covenant or agreement does not specify a term, until sixty (60) days following expiration of the applicable statute of limitations (giving effect to any extensions and waivers thereof. Notwithstanding the foregoing, any representation, warranty or covenant in respect of which indemnity may be sought under this Agreement prior to the Survival Date will survive the time at which it would otherwise terminate pursuant to the preceding sentences if written notice of the inaccuracy or breach thereof giving rise to such right of indemnity has been given to the party against whom indemnification may be sought prior to such time.

## 8.2 Indemnification.

- (a) Indemnification by the Seller Parties. Subject to the limitations expressly set forth in this ARTICLE VIII, from and after the Closing, Seller and the Principal Member will jointly and severally indemnify, defend and hold harmless Buyer and its Affiliates, and their respective and their respective members, shareholders, officers, directors, managers, employees, agents, successors and assigns (all such foregoing Persons, collectively, the "Buyer Indemnitees") from and against any Losses the Buyer Indemnitees may suffer, sustain or become subject to, arising out of, in connection with or resulting from:
  - (i) any breach or inaccuracy of any representation or warranty of Seller or the Principal Member contained in this Agreement, any other Transaction Document or in any certificate, instrument or document delivered by Seller or the Principal Member pursuant to this Agreement;
  - (ii) any breach of any covenant to be performed by Seller or the Principal Member pursuant to this Agreement, any other Transaction Document or in any certificate, instrument or document delivered by Seller or the Principal Member pursuant to this Agreement;
  - (iii) any Excluded Asset or Retained Liability, including but not limited to liabilities and obligations arising from ownership of the Acquired Assets or the operation of the Business (but not including Assumed Liabilities) on or prior to the Closing;

- (iv) the face amount of any Accounts Receivable that is not collected in full by the Buyer within 90 days after the due date of such Accounts Receivable; provided however that any uncollected Account Receivable will be assigned to the Seller within Thirty (30) days after Buyer exercises its indemnification right under this Section 8.2(a)(iv), and Seller may attempt to collect the same, and Buyer will provide the relevant documents to Seller to enable Seller to attempt to/collect the same; and
- (v) any items set forth on <u>Schedule 8.2(a)(v)</u>.
- (b) <u>Indemnification by Buyer</u>. Subject to the limitations expressly set forth in this ARTICLE VIII, from and after the Closing, Buyer will indemnify, defend and hold harmless Seller and its successors and assigns (all such foregoing persons, collectively, the "<u>Seller Indemnitees</u>") from and against any Losses Seller Indemnitees may suffer, sustain or become subject to, arising out of, in connection with or resulting from:
  - (i) any breach or inaccuracy of any representation or warranty of Buyer contained in this Agreement, any other Transaction Document or in any certificate, instrument or document delivered by Buyer pursuant to this Agreement, as of the date such representation or warranty was made or as if such representation or warranty was made on and as of the Closing (except for representations and warranties that expressly relate to a specified date, the inaccuracy in or breach of which will be determined with reference to such specified date);
  - (ii) any breach of any covenant to be performed by Buyer pursuant to this Agreement, any other Transaction Document or in any certificate, instrument or document delivered by Buyer pursuant to this Agreement; and
  - (iii) any Assumed Liability, Buyer's ownership of the Acquired Assets and Buyer's operation of the Business, including but not limited to liabilities and obligations arising from ownership of the Acquired Assets or the operation of the Business, in each case from and after the Closing.

## 8.3 Limitations on Liability

(a) Except with respect to any breaches of the Fundamental Representations or Section 3.8, with respect to Losses arising pursuant to Section 8.2(a)(i), the Buyer Indemnitees will not be entitled to indemnification under this Agreement until the aggregate amount of Losses for which the Buyer Indemnitees would be liable thereunder exceeds Forty Thousand Dollars (\$40,000.00)(the "Threshold Amount"), at which time, the Buyer Indemnitees will be entitled to indemnification for the aggregate amount of all Losses, including the Threshold Amount. Except in the case of fraud, intentional misrepresentation, or the breach of a Fundamental Representation, the aggregate maximum liability of the Seller Parties with respect to indemnification for Losses arising pursuant to Section 8.2(a)(i) shall not exceed an amount equal to \$750,000. Except in the case of fraud or intentional misrepresentation, the aggregate maximum liability of the Seller Parties with respect to indemnification for all Losses arising from breach of the Fundamental Representations shall not exceed an amount equal to the Purchase Price. Except as provided in the two immediately preceding sentences, neither Buyer, on the one hand, nor the Seller and the Principal Member, collectively, on the other hand, will be required to provide indemnification for aggregate Losses arising pursuant to Section 8.2(a)(i) (in the case of Seller and the Principal Member) or Section 8.2(b)(i) (in the case of Buyer) in excess of the Purchase Price.

(b) Each Indemnified Party (as defined below) shall use commercially reasonable efforts to make a claim for recovery of Losses that are recoverable by them under any insurance policy so long as such action does not result in a material premium increase for its insurance or negatively change the insurability of such Party's activities (whether in terms of coverable risks or otherwise), and the amount of an indemnifiable Loss shall be reduced to the extent such indemnifiable Loss is recovered by them under any such insurance policy. If any Buyer Indemnitee has been indemnified for a Loss hereunder and at any time thereafter such Buyer Indemnitee or any other Buyer Indemnitee recovers all or a portion of such Loss from the proceeds of an insurance policy, the Buyer Indemnitees shall promptly refund to the Escrow

Account (if during the term of the Escrow Account) or Seller (if the term of the Escrow Account has expired and the Escrow Account has been liquidated) the amount previously received by the Buyer Indemnitees (whether received from the Escrow Account or from or on behalf of Seller) with respect to such Loss (up to the amount recovered from such insurance policy). To the extent required by applicable law, each Indemnified Party hereunder shall take, and use commercially reasonably efforts to cause its Affiliates to take, commercially reasonable steps to mitigate any Losses upon becoming aware of any event or circumstance that would be reasonably expected to, or does, give rise thereto. No Indemnified Party shall be entitled to recover any amount due hereunder more than once in respect of the same Losses, including by reason of the state of facts giving rise to such Losses, constituting a breach of more than one representation, warranty or covenant.

#### 8.4 Claims Procedures.

(a) If a claim, action, suit or proceeding by a third party (a "Third Party Claim") is made against any Person entitled to indemnification pursuant to Section 8.2 hereof (an "Indemnified Party"), and if such Indemnified Party intends to seek indemnity with respect thereto under this ARTICLE VIII, such Indemnified Party shall promptly deliver a Claim Notice to the Party obligated to indemnify such Indemnified Party (such notified party, the "Responsible Party") of such claims; provided that the failure to deliver a Claim Notice shall not relieve the Responsible Party of its obligations hereunder, except to the extent that the Responsible Party is actually and materially prejudiced thereby. The Responsible Party shall have the right to be entitled to participate in the defense of such Third Party Claim and, to the extent that it wishes, to assume the conduct and control, through counsel reasonably acceptable to the Indemnified Party at the expense of the Responsible Party, of the settlement or defense thereof, provided that the Responsible Party notifies the Indemnified Party of its intent to exercise such right within 15 days after receipt of such Claim Notice, and upon receipt of such notice, the Indemnified Party shall reasonably cooperate with the Responsible Party in connection therewith; provided, that the Responsible Party shall not have the right to assume the defense of such Third Party Claim if (i) the Responsible Party fails to conduct the defense of the claim actively and diligently, (ii) any applicable limitation herein would serve to limit the obligation of the Responsible Party to indemnify the Indemnified Party for any Loss which would be reasonably anticipated to result from such claim were it successful or (iii) the Responsible Party is not reasonably anticipated to have the financial wherewithal to indemnify the Indemnified Party if the claim is successful. If the Responsible Party assumes the defense of a Third Party Claim, then (i) such assumption will conclusively establish for purposes of this Agreement that the claims made in that Third Party Claim are within the scope of and subject to indemnification, and (ii) no compromise or settlement of such Third Party Claims may be effected by the Responsible Party without the Indemnified Party's consent unless (A) there is no finding or admission of any violation of a Legal Requirement or any violation of the rights of any Person; (B) the sole relief provided is monetary damages that are paid in full by the Responsible Party; and (C) the Indemnified Party has no liability with respect to such compromise or settlement. The Responsible Party shall permit the Indemnified Party to participate in such settlement or defense through counsel chosen by such Indemnified Party, provided that the fees and expenses of such counsel shall be borne by such Indemnified Party. So long as the Responsible Party is reasonably contesting any such Third Party Claim in good faith, the Indemnified Party shall not pay or settle any such Third Party Claim. Notwithstanding the foregoing, the Indemnified Party shall have the right to pay or settle any such Third Party Claim (whether or not appropriate notice has been given by Indemnified Party), provided that in such event it shall irrevocably waive any and all rights (including any rights of its Affiliates) to indemnification for such Third Party Claim unless the Responsible Party shall have expressly consented in advance in writing to such payment or settlement. If the Responsible Party does not notify the Indemnified Party within 15 days after the receipt of the Indemnified Party's Claim Notice hereunder that it elects to undertake the defense of the Third Party Claim described in such Claim Notice, the Indemnified Party shall have the right to (x) contest such Third Party Claim, and (y) with the prior written consent of the Responsible Party (which consent shall not be unreasonably withheld, conditioned or delayed), settle such Third Party Claim, and the Responsible Party shall have the right at any time to participate (including through counsel chosen by the Responsible Party) in the defense of any such Third Party Claim at its own expense. The Indemnified Party shall not waive any right to indemnity pursuant to this Agreement by contesting or settling any claim in accordance with the previous sentence. The Responsible Party shall not, except with the consent of the Indemnified Party (which consent shall not be unreasonably withheld, conditioned or delayed), enter into any settlement or consent to the entry of any judgment that does not include as an unconditional term thereof the giving by the Person or Persons asserting such claim to all Indemnified Parties of an unconditional release from all liability with respect to such claim.

- (b) Notwithstanding the foregoing, the Responsible Party will not be entitled to assume (or retain, as applicable) control of such defense if (i) the Indemnified Party concludes in good faith that, in light of any actual or potential conflict of interest, it would be inappropriate for legal counsel selected by the Responsible Party to represent the Indemnified Party, or (ii) the such claim for fails to vigorously and continuously prosecute or defend such claim in good faith or fails to begin such prosecution or defense in a timely manner.
- (c) All of the Parties shall cooperate in the defense or prosecution of any Third Party Claim in respect of which indemnity may be sought hereunder and shall furnish such records, information and testimony, and attend such conferences, discovery proceedings, hearings, trials and appeals, as may be reasonably requested in connection therewith; provided, however, that this Section 5.3(c) shall not apply to any pending or anticipated adversarial, judicial or arbitral dispute or proceeding between the Parties to this Agreement, which shall be governed by the applicable rules of discovery, evidence or procedure. With respect to any Third Party Claim subject to indemnification hereunder, the Parties agree to cooperate in such a manner as to preserve in full (to the extent possible) the confidentiality of all confidential information and the attorney-client and work-product privileges of the other Party.
- (d) A claim for indemnification for any matter not involving a Third Party Claim may be asserted by delivery of a Claim Notice to the party from whom indemnification is sought and, unless contested in accordance with this Agreement, shall be paid promptly after such notice. The party from whom indemnification is sought will have a period of fifteen (15) days within which to respond in writing to such direct claim.
- (e) The Parties shall use commercially reasonable efforts with respect to any information shared pursuant to this <u>Section 8.4</u> to preserve attorney-client privilege.

8.5 **Payment of Claim**. Notwithstanding anything to the contrary in this Agreement and subject to the qualifications set forth in ARTICLE VIII, the obligation of Buyer to pay the Earnout Payments will be qualified in its entirety by the right of Buyer to reduce the amount of the Earnout Payments by the amount of indemnifiable Losses in respect of one or more indemnification claims to which any Buyer Indemnitee is entitled pursuant to ARTICLE VIII. Subject to the limitations set forth herein, in order to satisfy any indemnification obligations of any Seller Party with respect to any claim for indemnification pursuant to this ARTICLE VIII, a Buyer Indemnitee will have the right to recover indemnifiable Losses that have been incurred (i) first, from the Escrow Amount pursuant to the terms and conditions of the Escrow Agreement, (ii) second, from any Earnout Payments then due and payable, as determined in the sole and absolute discretion of the Buyer Indemnitee, and (iii) thereafter, from Seller and the Principal Member within three business days after demand is made by a Buyer Indemnitee. Notwithstanding the foregoing, Seller and the Principal Member will pay to Buyer all indemnifiable Losses in connection with Section 8.2(a)(iv) within three business days after demand is made by a Buyer Indemnitee. Subject to the limitations set forth herein, in order to satisfy any indemnification obligations of any Buyer Party with respect to any claim for indemnification pursuant to this ARTICLE VIII, a Seller Indemnitee will have the right to recover indemnifiable Losses that have been incurred from Buyer Indemnitee.

8.6 **Exclusive Remedy**. From and after the Closing, , except for a Party's right to specific performance or injunctive relief under Section 9.11, as provided in Section 2.6 or with respect to any fraud of any Party, the sole and exclusive remedy of any Indemnified Party with respect to any and all Losses arising in connection with the representation, warranties and covenants set forth in this Agreement will be pursuant to the indemnification obligations set forth in this ARTICLE VIII.

8.7 **Calculation of Loss**. As it relates to indemnification claims under this ARTICLE VIII, for purposes of determining whether a breach has occurred and calculating the amount of Losses incurred by a party seeking indemnification hereunder arising out of or resulting from any breach of a representation, warranty, covenant or agreement contained herein, references to "Material Adverse Effect" or materiality (or other similar terms) shall be disregarded. As used herein, Losses of a Person are not limited to matters asserted by Third Parties, but include Losses incurred or sustained by such Person in the absence of claims by Third Parties.

8.8 **Investigation**. The right to indemnification or any other remedy based on representations, warranties, covenants and agreements of Seller and the Principal Member in this Agreement, or any document, certificate or other instrument required to be delivered by Seller or the Principal Member under this Agreement will not be affected by any investigation conducted by any Indemnified Party or any other Person at any time, or any knowledge acquired (or capable of being acquired) by any Indemnified Party or any other

Person at any time, whether before or after the execution and delivery of this Agreement and prior to the Closing, with respect to the accuracy or inaccuracy of, or compliance with, any such representation, warranty, covenant or agreement.

8.9 Treatment of Indemnity Payments. Any amount paid for indemnification under this Article VIII will be treated as an adjustment to the Purchase Price, including for all Tax purposes, except as otherwise required by any Legal Requirement.

8.10 Schedules. All representations and warranties concerning the Business are qualified to the extent of any applicable disclosure in the Schedules making specific reference to the Section or Sections of this Agreement to which such disclosure applies, and such disclosure shall further qualify any other Section or Sections of this Agreement to the extent it is reasonably apparent on the face of such disclosure that such disclosure is intended to qualify any other Section of this Agreement.

-43-

# ARTICLE IX. MISCELLANEOUS

9.1 Notices. All notices, requests, consents, claims, demands, waivers and other communications hereunder shall be in writing and shall be deemed to have been given (a) when delivered, if personally delivered; (b) on the next Business Day after dispatch, if sent postage pre-paid by nationally recognized, overnight courier guaranteeing next business day delivery; (c) if sent by e-mail of a PDF document, the date when sent by email sent to the email address for the sender stated in this Section 9.1 (provided that, unless receipt of such email is acknowledged, such email is followed up within one Business Day by dispatch pursuant to one of the other methods described herein), or (d) on the third day after the date mailed, by certified or registered mail, return receipt requested, postage prepaid. Such communications must be sent to the respective Parties at the following addresses (or at such other address for a Party as shall be specified in a notice given in accordance with this Section 9.1):

If to a Buyer Party:

Zomedica Corp. 100 Phoenix Drive, Suite 125 Ann Arbor, Michigan 48108 Attention: Chief Executive Officer and General Counsel E-Mail: lheaton@zomedica.com and kdehaanfullerton@zomedica.com

with a copy (which shall not constitute notice to the Buyer Party) to:

Maslon LLP 90 South 7th Street, Suite 3300 Minneapolis, MN 55402 Attention: Alan M. Gilbert E-mail address: alan.gilbert@maslon.com

If to Seller:

Revo Squared LLC 2060 Franklin Way Suite 100 Marietta, GA 30067 Attention: William E. Campbell Email: bcampbell@duoview.net

with a copy (which shall not constitute notice to the Buyer Party) to:

Briskin, Cross & Sanford, LLC 33 South Main Street, Suite 300 Alpharetta, GA 30009 Attention: Alan M. Briskin

#### Email: abriskin@briskinlaw.com

9.2 **Computation of Time**. Whenever the last day for the exercise of any privilege or the discharge or any duty hereunder will fall upon a Saturday, Sunday, or any date on which banks in New York City, New York are authorized to be closed, the Party having such privilege or duty may exercise such privilege or discharge such duty on the next succeeding day which is a regular business day.

9.3 **Expenses**. Except as otherwise provided in this Agreement, all fees and expenses incurred in connection with this Agreement and the Contemplated Transactions, including the fees and disbursements of counsel, financial advisors and accountants, shall be paid by the Party incurring such fees or expenses.

-44-

9.4 **Governing Law; Jurisdiction**. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the law of any jurisdiction other than the State of Delaware. Except for the matters to be determined by the Neutral Accountant pursuant to Section 2.6 or Section 2.9, each of the Parties submits to the exclusive jurisdiction of any state or federal court within the State of Delaware in any action or proceeding arising out of or relating to this Agreement and agrees that all claims in respect of the action or proceeding shall be exclusively heard and determined in any such court. The Parties hereby irrevocably waive, to the fullest extent permitted by Applicable Law, any objection which they may now or hereafter have to the laying of venue of any such dispute brought in such court. Each of the Parties waives any defense of inconvenient forum to the maintenance of any action or proceeding so brought

9.5 **Parties in Interest**. This Agreement shall be binding upon and inure solely to the benefit of each Party and its successors and permitted assigns and, except as set forth in Section 8.2, nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person any rights, benefits or remedies of any nature whatsoever under or by reason of this Agreement.

9.6**Counterparts**. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same agreement. Delivery of an executed counterpart of a signature page to this Agreement by facsimile, portable document format or other electronic means shall be effective as delivery of a manually executed counterpart to this Agreement.

9.7Entire Agreement; Amendment; Waiver; Assignment. This Agreement and the Disclosure Schedule, other Schedules, Exhibits and other documents delivered pursuant to this Agreement (a) constitute the entire agreement among the Parties with respect to the subject matter hereof and thereof and supersedes all other prior agreements and understandings, both written and oral, among the Parties with respect to the subject matter hereof, (b) can be amended, supplemented or changed, and any provision hereof can be waived, only by written instrument making specific reference to this Agreement signed by the Buyer Parties, in the case of an amendment, supplement, modification or waiver sought to be enforced against the Buyer Parties, or Seller, in the case of an amendment, supplement, modification or waiver sought to be enforced against Seller, and (c) shall not be assigned by any Party (whether by operation of law or otherwise) without the prior written consent of the other Parties.

9.8**Severability**. If any term or other provision of this Agreement is invalid, illegal or unenforceable, all other provisions of this Agreement shall remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any Party.

9.9No Strict Construction. Each Party acknowledges that this Agreement has been prepared jointly by the Parties, and will not be strictly construed against any Party. As a consequence, the Parties do not intend that the presumptions of any Legal Requirements or rules relating to the interpretation of Contracts against the drafter of any particular clause should be applied to this Agreement and therefore waive the benefits of such construction.

9.10Waiver of Jury Trial. EACH OF THE PARTIES HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION, OR CAUSE OF ACTION (I) ARISING UNDER THIS AGREEMENT OR (II) IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES IN RESPECT OF THIS AGREEMENT OR ANY OF THE TRANSACTIONS RELATED HERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER IN CONTRACT, TORT, EQUITY, OR OTHERWISE. THE PARTIES EACH HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION, OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY AND THAT THE PARTIES MAY FILE AN ORIGINAL COUNTERPART OF A COPY OF THIS AGREEMENT WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE PARTIES TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

9.11 **Specific Performance**. Each Party acknowledges and agrees that the other Party would be damaged irreparably in the event any provision of this Agreement is not performed in accordance with its specific terms or otherwise is breached, so that a Party shall be entitled to seek injunctive relief to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in addition to any other remedy to which such Party may be entitled, at law or in equity. No limitation herein shall restrict any Party from seeking and obtaining equitable relief.

9.12 Failure or Indulgence not Waiver. No failure or delay of any Party in the exercise of any right hereunder will impair such right or be construed to be waiver of, or acquiescence in, any breach of any representation, warranty or agreement herein, nor will any exercise of any such right preclude any other or further exercise thereof or any other right.

\* \* \* \* \* \* \*

-46-

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed as of the day and year first above written.

**SELLER:** 

**BUYER:** 

# ZOMEDICA INC.

By: <u>/s/ Larry Heaton</u> Name: Larry Heaton Title: Chief Executive Officer

## **BUYER PARENT:**

#### ZOMEDICA CORP.

By: <u>/s/ Larry Heaton</u> Name: Larry Heaton Title: Chief Executive Officer

#### **REVO SQUARED LLC**

By: <u>/s/ William E. Campbell</u> Name: William E. Campbell Title: Chief Executive Officer

## **PRINCIPAL MEMBER:**

By: <u>/s/ William E. Campbell</u> Name: William E. Campbell

[Signature Page to Asset Purchase Agreement]



HOLD FOR RELEASE UNTIL Wednesday June 15<sup>th</sup>

# ZOMEDICA TO ACQUIRE ASSETS OF INNOVATIVE VETERINARY IMAGING COMPANY REVO SQUARED

**ANN ARBOR, Mich., June 15, 2022** (Accesswire) – Zomedica Corp. (NYSE American: ZOM) ("Zomedica" or the "Company"), a veterinary health company offering diagnostic and therapeutic products for companion animals, today announced it has signed a definitive agreement to acquire substantially all of the assets of Revo Squared LLC ("Revo").

Revo's innovative MicroView digital cytology platform will provide veterinarians with best-in-class image quality while providing the first in-clinic automated slide preparation system. Slide preparation is a critical part of cytology, and slide preparation errors are the number one reason cited by pathologists for being unable to read a slide. The MicroView platform improves practice workflow efficiency by automating the process to assure a consistent smear and stain in order to produce a readable slide.

Bill Campell, Chief Executive Officer of Revo Squared stated "We are excited to join the team at Zomedica; this deal will bring our MicroView product offering into the forefront of veterinary medicine. For the first time clinicians will be able to have hematology and cytology slides automatically smeared and stained inside a microscope, eliminating human error in slide preparation. Implementing MicroView in veterinary hospitals enhances diagnostics and reduces the wait time."

"We're excited to work with Bill and his colleague Negin Ghassemian as they join Zomedica as Vice President, Imaging Systems and Senior Director, Product Development," said Zomedica Chief Executive Officer, Larry Heaton, "They have done a great job of developing the MicroView system to this point, and we look forward to them leading the effort to integrate the MicroView system into the MyZomedica customer interface, commence manufacturing at our Zomedica manufacturing center in Alpharetta and bring it to market."

The Microview platform is intended for veterinary clinics of any size, especially those where a dedicated pathologist is not available on-site. The MicroView system allows veterinarians the option of reading the slide, and making a diagnosis themselves, or sending it digitally to the MicroView pathologist team for an expert second opinion.

"We're excited to reach a definitive agreement with Revo Squared and intend to close the transaction within 30 days." said Zomedica Vice President of Business Development and Strategic Planning, Greg Blair, "This furthers Zomedica's commitment to bring innovative technologies and tools to market that improve pet outcomes and pet parent satisfaction while also improving veterinary practice workflow efficiency and economics."

### About Revo Squared

Based in Marietta, Georgia, Revo Squared has long been an innovator in veterinary imaging systems including Digital Cytology, ultrasound, and digital x-ray systems.

### About Zomedica

Based in Ann Arbor, Michigan, Zomedica (NYSE American: ZOM) is a veterinary health company creating products for companion animals by focusing on the unmet needs of clinical veterinarians. Zomedica's product portfolio will include innovative diagnostics and medical devices that emphasize patient health and practice health. It is Zomedica's mission to provide veterinarians the opportunity to increase productivity and grow revenue while better serving the animals in their care. For more information, visit www.ZOMEDICA.com.

1

#### **Follow Zomedica**

- Email Alerts: http://investors.zomedica.com
- LinkedIn: https://www.linkedin.com/company/zomedica
- Facebook: https://m.facebook.com/zomedica
- Twitter: https://twitter.com/zomedica
- Instagram: https://www.instagram.com/zomedica\_inc

### **Reader Advisory**

Except for statements of historical fact, this news release contains certain "forward-looking information" or "forward-looking statements" (collectively, "forward-looking information") within the meaning of applicable securities law. Forward-looking information is frequently characterized by words such as "plan", "expect", "project", "intend", "believe", "anticipate", "estimate" and other similar words, or statements that certain events or conditions "may" or "will" occur and include statements relating to our expectations regarding future results. Although we believe that the expectations reflected in the forward-looking information are reasonable, there can be no assurance that such expectations will prove to be correct. We cannot guarantee future results, performance or achievements. Consequently, there is no representation that the actual results achieved will be the same, in whole or in part, as those set out in the forward-looking information.

Forward-looking information is based on the opinions and estimates of management at the date the statements are made, including assumptions with respect to American economic growth, demand for the Company's products, the Company's ability to produce and sell its products, our ability to successfully integrate and operate the PulseVet business, the sufficiency of our budgeted capital and operating expenditures, the cost, adequacy and availability of supplies required for our operations, the satisfaction by our strategic partners of their obligations under our commercial agreements, our ability to realize upon our business plans and cost control efforts and the impact of COVID-19 on our business, results, and financial condition.

Our forward-looking information is subject to a variety of risks and uncertainties and other factors that could cause actual events or results to differ materially from those anticipated in the forward-looking information. Some of the risks and other factors that could cause the results to differ materially from those expressed in the forward-looking information include, but are not limited to: uncertainty as to whether our strategies and business plans will yield the expected benefits; uncertainty as to the timing and results of development work and verification and validation studies; uncertainty as to the timing and results of commercialization efforts, as well as the cost of commercialization efforts, including the cost to develop an internal sales force and manage our growth; uncertainty as to our ability to successfully integrate and operate the Pulse Vet business, uncertainty as to our ability to supply equipment and assays in response to customer demand; uncertainty regarding the cost, adequacy and availability of supplies required for our operations; uncertainty as to the likelihood and timing of any required regulatory approvals, and the availability and cost of capital; the ability to identify, develop and/or achieve commercial success for new products and technologies; veterinary acceptance of our products; competition from related products; the level of expenditures necessary to maintain and improve the quality of products and services; changes in technology and changes in laws and regulations; our ability to secure and maintain strategic relationships; performance by our strategic partners of their obligations under our commercial agreements, including product manufacturing obligations: risks pertaining to permits and licensing, intellectual property infringement risks, risks relating to any required clinical trials and regulatory approvals, risks relating to the safety and efficacy of our products, the use of our products, intellectual property protection, risks related to the COVID-19 pandemic and its impact upon our business operations generally, including our ability to develop and commercialize our products, and the other risk factors disclosed in our filings with the SEC and under our profile on SEDAR at www.sedar.com. Readers are cautioned that this list of risk factors should not be construed as exhaustive.

2

The forward-looking information contained in this news release is expressly qualified by this cautionary statement. We undertake no duty to update any of the forward-looking information to conform such information to actual results or to changes in our expectations except as otherwise required by applicable securities legislation. Readers are cautioned not to place undue reliance on forward-looking information.

## Investor Relations Contact: PCG Advisory Group Kirin Smith, President ksmith@pcgadvisory.com +1.646.823.8656

Cover	Jun. 14, 2022
Cover [Abstract]	
Entity Registrant Name	ZOMEDICA CORP.
Entity Central Index Key	0001684144
Document Type	8-K
Amendment Flag	false
Entity Emerging Growth Company	false
Document Period End Date	Jun. 14, 2022
Entity Incorporation State Country Code Z4	
Written Communications	false
Soliciting Material	false
Pre Commencement Tender Offer	false
Pre Commencement Issuer Tender Offer	<u>r</u> false
Entity File Number	001-38298
Entity Address Address Line 1	100 Phoenix Drive
Entity Address Address Line 2	Suite 125
Entity Address City Or Town	Ann Arbor
Entity Address State Or Province	MI
Entity Address Postal Zip Code	48108
City Area Code	734
Local Phone Number	369-2555
Security 12b Title	Common Shares, without par value
Trading Symbol	ZOM
Security Exchange Name	NYSEAMER

Copyright © 2022 www.accidentes.com, All Rights Reserved. Please Consider the Environment Bettre Plinting This Document

l, "localname": "CityAreaCode", "nauri": "http://kbl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" l, "xkbltype": "normalizedStringItemType" "wbritype": "normalizedStringItemTyy
"del\_CoverAbstract": {
 "auth\_ref"; [],
 "lang": {
 "en-us"; {
 "cloa": {
 "cloa": {
 "documentation": "Cover page.",
 "label": "Cover [Abstract]"
 }
} ), "localname": "CoverAbstract", "nsuri": "http://xbrl.sec.gov/dei/2022", "xbrltype": "stringItemType" 1. ], "auth\_reft: [], "auth\_reft: [], "augh\_reft: [], "a

}, "localname": "AmendmentFlag", "nsuri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" !,
"xbrltype": "booleanItemType" ADDitype : Doubletinge fai\_titype.code"; { "auth\_raf": [], "ang"; [] "en-us": [ "documentation": "Area code of city", "label": "City Area Code"

l, "localname": "DocumentPeriodEndDate", "nsurf: "http://kbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" ], "kbritype": "dateItenType" ], ), "dei\_DocumentType": { "auth\_ref": [],

nds previously-filed or accepted submission."

"htal" "passboft "zom \$k.htm", "contextBaft "From202-06-14t02022-06-14t", "desimal" unl, "first" trop, "ang", "desimal typegistrantName", "reportCount" 1, "anigue", trop, "an

"
ttp://wil.ac.gov/dd/2022\*: 2,
if.if.ac.gov/dd/2022\*: 2,
if.ac.gov/dd/2022\*: 2,
if.ac.gov/dd/202\*: 2,
i 

]/ \*elementCount": 23, "entityCount": 1, "hidden": { "hitge://xbrl.sec.gov/dei/2022": 2, "total": 2 },

};
"presentationLink": {
 "local": [
 "zom-20220614\_pre.xml"
} 'schema": ( "local": [ "zom-20220614.xsd" -loc1: [
 tan=202061.usd"
 tan=202061.usd"
 tan=202061.usd"
 tan=202061.usd"
 tan=202061.usd"
 tan=202071.usd"
 tan=2020

), "definitionLink": ( "local": [ "zom-20220614\_def.xml" },
"inline": (
 "local": [
 "zom\_8k.htm"
]

[ "instance": {
 "zom\_8k.htm": {
 "axisCustom": 0,
 "axisStandard": 0,
 "contextCount": 1,
 "dta": 1,
 "cloulationLink": {
 "localuiationLink": {
 "localu: {
 "zom=20220614\_cal.xml"
 ]
}

), "labelLink": { "local": [ "zom-20220614\_lab.xml" ]

Copyright © 2022 www.accidentes.com, All Rights Reserved. Please Consider the Environment Bettre Plinting This Document

suffix may contain 1-4 chara

rs, and the fields are separated with a hy

}, "localname": "EntityFileNumber", "nsuri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover"] ], "xbrltype": "fileNumberItemType" }, "localname": "EntityIncorporationStateCountryCo "neuri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover"]; 1,
"xbrltype": "edgarStateCountryItenType" 'dei EntityRegistrantName": { "auth\_ref": { "r1" }, "localname": "EntityRegistrantName", "nauri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" ]. l,
"xbrltype": "normalizedStringItemType" 

}, "localname": "LocalPhoneNumber", "nsuri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover"

'' 'localname": "EntityAddressAddressLine2", "nauri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" ], 1,
"xbrltype": "normalizedStringItemType" "bbitype": "normalisedtringitestype" "auth.ret">(). "auth.ret">(). "auth.ret">(). "augit ( "enup")( "documentation": "Name of the City or Town", "label": "Entity Address City Or Town", ), "localname": "EntityAddressCityOrTown", "mswrit": "http://kbil.sec.gov/dei/2022\*, "presentation": [ "http://zomedica.com/role/Cover" ], "abityppe": "normalizedStringItemType" monttype: "normalized tingitarype"
"add for type: "normalized tingitarype"
"auth ref" [],
"auth ref" [],
"ang": [
"arous": [
"cole": [
add tingitary "Code for the postal or tip code",
"about" "Entity Address Postal Sip Code" ), "localname": "EntityAddressPostalZipCode", "nauri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" 1,
"xbrltype": "normalizedStringItemType" }, "localname": "EntityAddressStateOrProvince", "nsuri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" !,
"xbrltype": "stateOrProvinceItemType" ), "dei\_EntityCentralIndexKey": { "auth\_ref": [ "r1" l. "ind": [ "de-us": [ "rola": [ "documentation": "A unique 10-digit SEC-issued value to identify entities that have filed disclosures with the SEC. It is commonly abbre "isbal": "Entity Control Tobes Kay" },
"localname": "EntityCentralIndexKey",
"nsuri": "http://xbrl.sec.gov/dei/2022",
"presentation": [
"http://zomedica.com/role/Cover" 1,
"xbrltype": "centralIndexKeyItemType" },
"dei\_EntityEmergingGrowthCompany": {
 "auth\_ref": [
 "z1" "L1 "indo": [ "en-us": [ "en-us": [ "en-us": [ "indowstation": "Indicate if registrant meets the emerging growth company criteria.", "indowstation": "Entity Emerging Growth Company" ] ) "localname": "EntityEmergingGrowthCompany", "mauri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" ) 1,
"xbrltype": "booleanItemType" in 1-3 digits, the se "miriyes" "limemetricempys" "dai\_EntityhonerainditatoContryCole": { "lash" [], "sash", "security [], "security [], "security [], "socenatization", "Non-character ECOM code representing the state or country of incorporation.", "saba", "Birly Incorporation ECOM code" 

etc). The document type is limited to the

rting SEC submission type, or the word 'Other'."

). "An EntryAdressAddressIng"; [ "Lag"; [] "Lag"; [] "Lag"; [] "Address Lins | such as Atts, Building Hame, Street Hame", "Lagd"; "Building Hame, Street Hame", "Lagd"; "Lagd; "Lagd"; "Lagd"; "Lagd"; "Lagd; "Lagd"; "Lagd; "Lagd"; "Lagd; }, "localname": "EntityAddressAddressLinel", "nauri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover"] !,
"xbrltype": "normalizedStringItemType"

',
'localname": "DocumentType",
"nauri": "http://xbrl.sec.gov/dei/2022",
"presentation": [
"http://zomedica.com/role/Cover"
1. ], "xbrltype": "submissionTypeItemType"

], "xbrltype": "normalizedStringItemType" ), "dei\_PreCommencementIssuerTenderOffer": { "auth\_ref": [ "r3"

"rr" "inag": [ "Henuar": [ "Accurate for "Boolsan flag that is true when the Form B-K filing is intended to satisfy the filing obligation of the registrant as pre-commencement communications pursuant to Bule 18-4(c) under the Exchange Act.", "Accurate for the formation of the registrant as pre-commencement communications pursuant to Bule 18-4(c) under the Exchange Act.", "Stabult" "Fre Commencement Issuer Tender Offer"

) "localname": "PreCommencementIssuerTenderOffer", "nsurl": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" ]. 1, "xbrltype": "booleanItemType"

mement communications pursuant to Rule 14d-2(b) under the Exchange Act.",

i."iccalname": "PreCommencementTenderOffer", "nsur!": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://romedica.com/role/Cover" ], "xbrltype": "booleanItenType" ],

i,
 "dei\_Securityl2bTitle": {
 "auth\_ref": [
 "r0"
].

"to" "anag": [ "anag": [ "sole": [ "documentation": "Title of a 12(b) registered security.", "documentation": "Security 12b Title" ]

"xbrltype": "securityTitleItemType"

i\_ i\_ccalname". "SecurityExchangeName", "neuri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://xomedica.com/role/Cover" ], "xbrltype": "edgarExchangeCodeItenType"

"dei\_SolicitingMaterial": (
"auth\_ref": [
"r5" ].

" "ino"; [ "an-asi; [ "cla"; [ "documentation": "Boolean flag that is true when the Form 8-K filing is intended to satisfy the filing obligation of the registrant as soliciting material pursuant to Rule 14a-12 under the Exchange Act.", "label": "Soliciting Material"

) "localname": "SolicitingNaterial", "nauri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" ], l, "xbrltype": "booleanItemType"

"metrys" "motokaaltemys" i\_\_\_\_\_\_\_ Traingsput "such\_ref" [], "such\_ref" [], "such\_ref" [], "such\_ref" [], "such\_ref" [], "such\_ref" [], "such\_ref", "Traing symbol of an instrument as listed on an exchange.", "subbul", "Training Symbol" |

) "localname": "TradingSymbol", "nauri": "http://xbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" 1.

l,
"xbrltype": "tradingSymbolItemType"

mainippe: "transpipe: transpipe: transp

], "localname": "WrittenCommunications", "nauris: "http://kbrl.sec.gov/dei/2022", "presentation": [ "http://zomedica.com/role/Cover" ], "xkpltype": "booleanItemType"

), "unitCount": 3

}
}
std ref": {
 "r0": {
 "xnme": "Exchange Act",
 "Number": "240",
 "Publisher": "SEC",
 "Section": "12",
 "Subsection": "12",
}

},
"rl":{
"Name": "Exchange Act",
"Number": "240",
"Publisher": "3EC",
"Section": "12",
"Subsection": "b-2"

},
"r2":{
"Name": "Exchange Act",
"Number": "240",
"Publisher": "SEC",
"Section": "12",
"Subsection": "d1-1"
},

},
"r3": {
"Name": "Exchange Act",
"Number": "240",
"Publisher": "25C",
"Section": "13e",
"Subsection": "4c"
1.

}, "z4": { "Name": "Exchange Act", "Number": "240", "Publisher": "25C", "Subsection": "14d", "Subsection": "2b" 1.

}, "S": { "Name": "Exchange Act", "Number": "240", "Publisher": "SEC", "Section": "14a", "Subsection": "12" ".

"r6": {
 "Name": "Securities Act",
 "Number": "230",
 "Publisher": "SEC",
 "Section": "425"
}

), "version": "2.1"